

IMPORTANT NOTICE REGARDING COVID 19 AND PARTICIPATION IN THE PERSONNEL AND ADMINISTRATION COMMITTEE MEETING

The Personnel and Administration Committee meeting will be held by teleconference in order to minimize the spread of the COVID 19 virus, in accordance with the State of Emergency proclaimed by Governor Newsom on March 4, 2020, Executive Order N 29 20 issued by Governor Newsom on March 17, 2020, and the Shelter in Place Order issued by the Monterey County Health Officer on March 17, 2020, as may be periodically amended.

To participate in this Personnel and Administration Committee meeting, the public is invited to observe and address the Committee telephonically or electronically. Instructions for public participation are below:

 For ZOOM participation please join by computer audio at: https://montereycty.zoom.us/j/91370408511
OR to participate by phone call any of these numbers below: +1 669 900 6833 US (San Jose)
+1 346 248 7799 US (San Jose)
+1 312 626 6799 US (Houston)
+1 312 626 6799 US (Chicago)
+1 929 205 6099 US (New York)
+1 253 215 8782 US
+1 301 715 8592 US

Enter this Meeting ID number: 913 7040 8511 PASSWORD: 420486 when prompted. Please note there is no Participant Code, you will just hit # again after the recording prompts you.

You will be placed in the meeting as an attendee; when you are ready to make a public comment, if joined by computer audio, please Raise your Hand; and by phone, please push *9 on your keypad.

2. If you wish to comment on a specific agenda item while the matter is being heard, you may participate by the following means:

When the Chair calls for public comment on an agenda item, the Zoom Meeting Host, or his or her designee, will first ascertain who wants to comment (among those who are in the meeting electronically or telephonically) and will then call on speakers and unmute their device one at a time. Public speakers may be broadcast in audio form only.

3. If you wish to comment on a particular agenda item, please submit your comments in writing via email to Monterey County Water Resources Agency at WRApubliccomment@co.monterey.ca.us by 5:00 p.m. on the Tuesday prior to the Committee meeting. To assist Agency staff in identifying the agenda item to which the comment relates please indicate the Personnel and Administration Committee meeting date and agenda number in the subject line. Comments received by the 5:00

p.m. Thursday deadline will be distributed to the Committee and will be placed in the record.

4. If you wish to make either a general public comment for items not on the day's agenda or to comment on a specific agenda item as it is being heard, please submit your comment, limited to 250 words or less, to the Monterey County Water Resources Agency at

WRApubliccomment@co.monterey.ca.us. In an effort to assist Agency staff in identifying the agenda item relating to your public comment please indicate in the subject line, the meeting body (i.e. Finance Committee) and item number (i.e. Item No. 10). Every effort will be made to read your comment into the record, but some comments may not be read due to time limitations. Comments received after an agenda item will be made part of the record if received prior to the end of the meeting.

5. If speakers or other members of the public have documents they wish to distribute to the Committee for an agenda item, they are encouraged to submit such documents by 5:00 p.m. on Tuesday before the meeting to: WRApubliccomment@co.monterey.ca.us. To assist Agency staff in identifying the agenda item to which the comment relates, the public is requested to indicate the Finance Committee date and agenda number in the subject line.

6. If members of the public want to present documents/Power Point presentations while speaking, they should submit the document electronically by 5:00 p.m. on Thursday before the meeting at WRApubliccomment@co.monterey.ca.us. (If submitted after that deadline, staff will make best efforts, but cannot guarantee, to make it available to present during the Committee meeting.)

7. Individuals with disabilities who desire to request a reasonable accommodation or modification to observe or participate in the meeting may make such request by sending an email to WRApubliccomment@co.monterey.ca.us. The request should be made no later than noon on the Thursday prior to the Committee meeting in order to provide time for the Agency to address the request.

8. The Chair and/or Secretary may set reasonable rules as needed to conduct the meeting in an orderly manner.

AVISO IMPORTANTE SOBRE COVID 19 Y PARTICIPACIÓN EN LA REUNIÓN DEL COMITE DE PERSONAL Y ADMINISTRACION

La reunión del Comité de Personal y Administracion se llevará a cabo por teleconferencia para minimizar la propagación del virus COVID 19, de acuerdo con el Estado de Emergencia proclamado por el Gobernador Newsom el 4 de Marzo del 2020, Orden Ejecutiva N 29 20 emitida por el Gobernador Newsom el 17 de Marzo del 2020, y la Orden de Refugio en el Lugar") emitida por el Oficial de Salud del Condado de Monterey el 17 de Marzo del 2020, según se pueda enmendar periódicamente. Para participar en esta reunión del Comité de Personal y Administracion el público están invitados a observar y dirigirse al Comité telefónicamente o por vía electrónica. Las instrucciones para la participación pública están a continuación: 1. El público puede observar la reunión ZOOM a través de computadora haciendo clic en el siguiente enlace: https://montereycty.zoom.us/j/91370408511 O el público puede escuchar a través del teléfono llamando al: +1 669 900 6833 US (San Jose) +1 346 248 7799 US (Houston) +1 312 626 6799 US (Chicago) +1 929 205 6099 US (New York) +1 253 215 8782 US +1 301 715 8592 US

Cuando se le solicite el código de acceso para entrar a la reunión, presione los siguientes números: 913 7040 8511 PASSWORD: 420486. Tenga en cuenta que no hay un Código de participante, simplemente presionará # nuevamente después de que la grabación lo solicite.

Se le colocará en la reunión como asistente; cuando esté listo para hacer un comentario público si se une al audio de la computadora, levante la mano; y por teléfono presione * 9 en su teclado.

2. Los miembros del público que desean comentar en un artículo específico de la agenda, mientras que el artículo se este presentando durante la reunión, pueden participar por cualquiera de los siguientes medios:

Cuando el Presidente del Comité solicite comentarios públicos sobre un artículo de la agenda, el anfitrión de la reunión Zoom o su designado, primero determinará quién quiere testificar (entre los que están en la reunión por vía electrónica o telefónica) y luego llamará a los oradores (speakers) y activará la bocina para el orador, uno a la vez. Todo orador, será transmitido por audio en altavoz solamente.

3. Si un miembro del público desea comentar sobre un artículo de la agenda en particular, se le es sumamente recomendable que envie sus comentarios por escrito por correo electrónico a la Agencia de Administración de Recursos del Agua (Agencia) a WRApubliccomment@co.monterey.ca.us antes de las 5:00 P. M. el Jueves antes de la reunión del Comité. Para ayudar al personal de la Agencia a identificar el número del artículo de la agenda con el cual se relaciona el comentario, se solicita al público que indique la fecha de la reunión del Comité y el número del artículo de la agenda en la línea de asunto. Comentarios recibidos en la fecha limite del Jueves a las 5 P.M, serán distribuidos al Comité y serán colocados en el registro.

4. Los miembros del público que deseen hacer un comentario público general para temas que no están en la agenda del día o que deseen comentar en un artículo específico mientras se escucha la

presentación, lo pueden hacer enviando un comentario por correo electrónico, preferiblemente limitado a 250 palabras o menos, a WRApubliccomment@co.monterey.ca.us. Para ayudar al personal de la Agencia a identificar el artículo de la agenda con el cual se relaciona el comentario, se solicita al público que indique el nombre del Comité (por ejemplo: Comité de Personal y Administracion) y el número del artículo de la agenda (por ejemplo: Artículo # 10). Se hará todo lo posible para leer el comentario en el registro, pero algunos comentarios pueden no leerse en voz alta debido a limitaciones de tiempo. Los comentarios recibidos después del cierre del período de comentarios públicos sobre un artículo de la agenda serán parte del registro si se reciben antes que termine la reunión del Comité.

5. Si los oradores u otros miembros del público tienen documentos que desean distribuir al Comité para un artículo de la agenda, se les recomienda enviar dichos documentos antes de las 5:00 P.M. el Jueves antes de la reunión a: WRApubliccomment@co.monterey.ca.us. Para ayudar al personal de la Agencia a identificar el número del artículo de la agenda con el cual se relaciona el comentario, se solicita al público que indique la fecha de la reunion del Comité y el número de agenda en la línea de asunto.

6. Si los miembros del público desean presentar documentos o presentaciones de PowerPoint mientras hablan, deben enviar el documento electrónicamente antes de las 5:00 P.M. del Jueves antes de la reunión a WRApubliccomment@co.monterey.ca.us (Si se presenta después de ese plazo, el personal hará los mejores esfuerzos, pero no puede garantizar que esté disponible su PowerPoint para presentar durante la reunión del Comité).

7. Las personas con discapacidades que deseen solicitar una modificación o modificación razonable para observar o participar en la reunión pueden realizar dicha solicitud enviando un correo electrónico a WRApubliccomment@co.monterey.ca.us. La solicitud debe hacerse a más tardar el mediodía del Jueves antes de a la reunión del Comité para dar tiempo a la Agencia para que atienda la solicitud.

8. El Presidente y / o Secretario pueden establecer reglas razonables según sea necesario para llevar a cabo la reunión de manera ordenada.

Call to Order

<u>Roll Call</u>

Public Comment

Consent Calendar

1.

Approve the Minutes of the Personnel and Administration Committee meeting held on May 7, 2021.

Attachments: Draft Action Minutes May 7, 2021

Scheduled Items

2		

Consider supporting a recommendation that the Monterey County Water Resources Agency Board of Supervisors:

a. Approve Amendment No. 3 to the Amended and Restated Water Recycling Agreement with Monterey One Water for a time extension related to the Conditions Precedent for the New Source Water Facilities; and

b. Request the Chair of the Monterey County Water Resources Agency Board of Supervisors to execute the Amendment.

Attachments: **Board Report** ARWRA Amendment No. 3 Amendment No. 2 Amendment No. 1 Amended & Restated Water Recycling Agreement

Status Reports

3.	Receive an update on personnel activities	
4.	Receive an update on the Training and Professional Growth Plan	
5.	Receive an update on real property and Agency land appraisals	
	Attachments: Land Use and Lease project list June 2021	
Calendar		

6.	Set next meeting date and discus	ss future agenda items
----	----------------------------------	------------------------

Adjournment

Monterey County

Board Report

Legistar File Number: WRAPA 21-021

Introduced: 5/27/2021

Version: 1

Approve the Minutes of the Personnel and Administration Committee meeting held on May 7, 2021.

Monterey County



Item No.1 Board of Supervisors Chambers

Chambers 168 W. Alisal St., 1st Floor Salinas, CA 93901

June 04, 2021

Current Status: Draft Matter Type: WRA P&A Item

Page 1

PA 21-021



IMPORTANT NOTICE REGARDING COVID 19 AND PARTICIPATION IN THE PERSONNEL AND ADMINISTRATION COMMITTEE MEETING

The Personnel and Administration Committee meeting will be held by teleconference in order to minimize the spread of the COVID 19 virus, in accordance with the State of Emergency proclaimed by Governor Newsom on March 4, 2020, Executive Order N 29 20 issued by Governor Newsom on March 17, 2020, and the Shelter in Place Order issued by the Monterey County Health Officer on March 17, 2020, as may be periodically amended.

To participate in this Personnel and Administration Committee meeting, the public is invited to observe and address the Committee telephonically or electronically. Instructions for public participation are below:

 For ZOOM participation please join by computer audio at: https://montereycty.zoom.us/j/91422870915
OR to participate by phone call any of these numbers below: +1 669 900 6833 US (San Jose)
+1 346 248 7799 US (Bouston)
+1 312 626 6799 US (Chicago)
+1 929 205 6099 US (New York)
+1 253 215 8782 US
+1 301 715 8592 US

Enter this Meeting ID number: 914 2287 0915 PASSWORD: 522970 when prompted. Please note there is no Participant Code, you will just hit # again after the recording prompts you.

You will be placed in the meeting as an attendee; when you are ready to make a public comment, if joined by computer audio, please Raise your Hand; and by phone, please push *9 on your keypad.

2. If you wish to comment on a specific agenda item while the matter is being heard, you may participate by the following means:

When the Chair calls for public comment on an agenda item, the Zoom Meeting Host, or his or her designee, will first ascertain who wants to comment (among those who are in the meeting electronically or telephonically) and will then call on speakers and unmute their device one at a time. Public speakers may be broadcast in audio form only.

3. If you wish to comment on a particular agenda item, please submit your comments in writing via email to Monterey County Water Resources Agency at WRApubliccomment@co.monterey.ca.us by 5:00 p.m. on the Tuesday prior to the Committee meeting. To assist Agency staff in identifying the agenda item to which the

comment relates please indicate the Personnel and Administration Committee meeting date and agenda number in the subject line. Comments received by the 5:00 p.m. Thursday deadline will be distributed to the Committee and will be placed in the record.

4. If you wish to make either a general public comment for items not on the day's agenda or to comment on a specific agenda item as it is being heard, please submit your comment, limited to 250 words or less, to the Monterey County Water Resources Agency at WRApubliccomment@co.monterey.ca.us. In an effort to assist Agency staff in identifying the agenda item relating to your public comment please indicate in the subject line, the meeting body (i.e. Finance Committee) and item number (i.e. Item No. 10). Every effort will be made to read your comment into the record, but some comments may not be read due to time limitations. Comments received after an agenda item will be made part of the record if received prior to the end of the meeting.

5. If speakers or other members of the public have documents they wish to distribute to the Committee for an agenda item, they are encouraged to submit such documents by 5:00 p.m. on Tuesday before the meeting to: WRApubliccomment@co.monterey.ca.us. To assist Agency staff in identifying the agenda item to which the comment relates, the public is requested to indicate the Finance Committee date and agenda number in the subject line.

6. If members of the public want to present documents/Power Point presentations while speaking, they should submit the document electronically by 5:00 p.m. on Thursday before the meeting at WRApubliccomment@co.monterey.ca.us. (If submitted after that deadline, staff will make best efforts, but cannot guarantee, to make it available to present during the Committee meeting.)

7. Individuals with disabilities who desire to request a reasonable accommodation or modification to observe or participate in the meeting may make such request by sending an email to WRApubliccomment@co.monterey.ca.us. The request should be made no later than noon on the Thursday prior to the Committee meeting in order to provide time for the Agency to address the request.

8. The Chair and/or Secretary may set reasonable rules as needed to conduct the meeting in an orderly manner.

AVISO IMPORTANTE SOBRE COVID 19 Y PARTICIPACIÓN EN LA REUNIÓN DEL COMITE DE PERSONAL Y ADMINISTRACION

La reunión del Comité de Personal y Administracion se llevará a cabo por teleconferencia para minimizar la propagación del virus COVID 19, de acuerdo con el Estado de Emergencia proclamado por el Gobernador Newsom el 4 de Marzo del 2020, Orden Ejecutiva N 29 20 emitida por el Gobernador Newsom el 17 de Marzo del 2020, y la Orden de Refugio en el Lugar") emitida por el Oficial de Salud del Condado de Monterey el 17 de Marzo del 2020, según se pueda enmendar periódicamente.

Para participar en esta reunión del Comité de Personal y Administracion el público están invitados a observar y dirigirse al Comité telefónicamente o por vía electrónica. Las instrucciones para la participación pública están a continuación:

 El público puede observar la reunión ZOOM a través de computadora haciendo clic en el siguiente enlace: https://montereycty.zoom.us/j/ 91422870915
O el público puede escuchar a través del teléfono llamando al: +1 669 900 6833 US (San Jose)
+1 346 248 7799 US (Houston)
+1 312 626 6799 US (Chicago)
+1 929 205 6099 US (New York)
+1 253 215 8782 US
+1 301 715 8592 US

Cuando se le solicite el código de acceso para entrar a la reunión, presione los siguientes números: 914 2287 0915 PASSWORD: 522970. Tenga en cuenta que no hay un Código de participante, simplemente presionará # nuevamente después de que la grabación lo solicite.

Se le colocará en la reunión como asistente; cuando esté listo para hacer un comentario público si se une al audio de la computadora, levante la mano; y por teléfono presione * 9 en su teclado.

2. Los miembros del público que desean comentar en un artículo específico de la agenda, mientras que el artículo se este presentando durante la reunión, pueden participar por cualquiera de los siguientes medios:

Cuando el Presidente del Comité solicite comentarios públicos sobre un artículo de la agenda, el anfitrión de la reunión Zoom o su designado, primero determinará quién quiere testificar (entre los que están en la reunión por vía electrónica o telefónica) y luego llamará a los oradores (speakers) y activará la bocina para el orador, uno a la vez. Todo orador, será transmitido por audio en altavoz solamente.

3. Si un miembro del público desea comentar sobre un artículo de la agenda en particular, se le es sumamente recomendable que envie sus comentarios por escrito por correo electrónico a la Agencia de Administración de Recursos del Agua (Agencia) a WRApubliccomment@co.monterey.ca.us antes de las 5:00 P. M. el Jueves antes de la reunión del Comité. Para ayudar al personal de la Agencia a identificar el número del artículo de la agenda con el cual se relaciona el comentario, se solicita al público que indique la fecha de la reunión del Comité y el número del artículo de la agenda en la línea de asunto. Comentarios recibidos en la fecha limite del Jueves a las 5 P.M, serán distribuidos al Comité y serán colocados en el registro.

4. Los miembros del público que deseen hacer un comentario público general para temas que no están en la agenda del día o que deseen comentar en un artículo específico mientras se escucha la presentación, lo pueden hacer enviando un comentario por correo electrónico, preferiblemente limitado a 250 palabras o menos, a WRApubliccomment@co.monterey.ca.us. Para ayudar al personal de la Agencia a identificar el artículo de la agenda con el cual se relaciona el comentario, se solicita al público que indique el nombre del Comité (por ejemplo: Comité de Personal y Administracion) y el número del artículo de la agenda (por ejemplo: Artículo # 10). Se hará todo lo posible para leer el comentario en el registro, pero algunos comentarios pueden no leerse en voz alta debido a limitaciones de tiempo. Los comentarios recibidos después del cierre del período de comentarios públicos sobre un artículo de la agenda serán parte del registro si se reciben antes que termine la reunión del Comité.

5. Si los oradores u otros miembros del público tienen documentos que desean distribuir al Comité para un artículo de la agenda, se les recomienda enviar dichos documentos antes de las 5:00 P.M. el Jueves antes de la reunión a: WRApubliccomment@co.monterey.ca.us. Para ayudar al personal de la Agencia a identificar el número del artículo de la agenda con el cual se relaciona el comentario, se solicita al público que indique la fecha de la reunion del Comité y el número de agenda en la línea de asunto.

6. Si los miembros del público desean presentar documentos o presentaciones de PowerPoint mientras hablan, deben enviar el documento electrónicamente antes de las 5:00 P.M. del Jueves antes de la reunión a WRApubliccomment@co.monterey.ca.us (Si se presenta después de ese plazo, el personal hará los mejores esfuerzos, pero no puede garantizar que esté disponible su PowerPoint para presentar durante la reunión del Comité).

7. Las personas con discapacidades que deseen solicitar una modificación o modificación razonable para observar o participar en la reunión pueden realizar dicha solicitud enviando un correo electrónico a WRApubliccomment@co.monterey.ca.us. La solicitud debe hacerse a más tardar el mediodía del Jueves antes de a la reunión del Comité para dar tiempo a la Agencia para que atienda la solicitud .

8. El Presidente y / o Secretario pueden establecer reglas razonables según sea necesario para llevar a cabo la reunión de manera ordenada.

Call to Order

The meeting was called to order at 8:30 a.m.

Roll Call

Present: Mark Gonzalez, Mike LeBarre, Jon Baillie and Matthew Simis

Absent: None

Public Comment

None

Consent Calendar

1. Approve the Minutes of the Personnel and Administration Committee meeting held on March 5, 2021.

Attachments: Draft Action MInutes March 5, 2021

Upon Motion by John Baillie and Second by Matthew Simis the Committee approved the Minutes of the Personnel and Administration Committee meeting held on March 5, 2021.

Ayes: Gonzalez, LeBarre, Baillie and Simis Noes: None

Status Reports

- **2.** Receive an update on personnel activities
- **3.** Receive an update on the Training and Professional Growth Plan
- 4. Receive an update on real property and Agency land appraisals

Attachments: Land Use and Lease project list May P&A

<u>Calendar</u>

5. Set next meeting date and discuss future agenda items

Adjournment

The meeting adjourned at 9:02 a.m.





Board Report

Legistar File Number: WRAPA 21-026

June 04, 2021

Item No.2

Board of Supervisors Chambers

168 W. Alisal St., 1st Floor Salinas, CA 93901

Introduced: 5/28/2021

Version: 1

Current Status: Agenda Ready Matter Type: WRA P&A Item

Consider supporting a recommendation that the Monterey County Water Resources Agency Board of Supervisors:

a. Approve Amendment No. 3 to the Amended and Restated Water Recycling Agreement with Monterey One Water for a time extension related to the Conditions Precedent for the New Source Water Facilities; and

b. Request the Chair of the Monterey County Water Resources Agency Board of Supervisors to execute the Amendment.

RECOMMENDATION:

It is recommended that the Personnel and Administration Committee support a recommendation by the Monterey County Water Resources Agency that the Monterey County Water Resources Agency Board of Supervisors:

a. Approve Amendment No. 3 to the Amended and Restated Water Recycling Agreement with Monterey One Water for a time extension related to the Conditions Precedent for the New Source Water Facilities; and

b. Request the Chair of the Monterey County Water Resources Agency Board of Supervisors to execute the Amendment.

SUMMARY/DISCUSSION:

The Monterey County Water Resources Agency (MCWRA) and Monterey One Water (M1W), formerly known as Monterey Regional Water Pollution Control Agency (MRWPCA), entered into an Amended and Restated Water Recycling Agreement (Restated Agreement), which included consideration of the financing, design, construction, operation, maintenance, and replacement of New Source Water Facilities to provide additional recycled water to MCWRA for use in the existing Castroville Seawater Intrusion Project (CSIP), a coastal irrigation project. In addition, M1W would be provided approximately 4,320 AFY of new source water to supplement the Pure Water Monterey Groundwater Replenishment Project which has been developed to provide drinking water.

Portions of the Amended and Restated Water Recycling Agreement applicable to the New Source Water Facilities will not become effective until the following conditions in Section 16.15 are all met:

1. Water Rights for the Blanco Drain and Reclamation Ditch are obtained from the California State Water Resources Control Board.

2. A fully executed, and California Public Utilities Commission approved, Water Purchase Agreement, between MRWPCA, MPWMD, and California-American Water, is approved by the California Public Utilities Commission and executed by the parties thereto. 3. Written finding by the Regional Water Quality Control Board that utilization of the Blanco Drain dry weather flows as New Source Water meets all treatment requirements for the aforesaid dry weather flows.

4. An independent third-party review of proposed capital and operating costs and preparation of an Engineer's Report is approved by the MCWRA Board of Directors and Board of Supervisors. The costs of the aforesaid third-party review shall be shared equally between MCWRA and MRWPCA.

5. A successful assessment or Proposition 218 process for rates and charges related to the operation and maintenance of the New Source Water Facilities and proportional primary and secondary treatment charges.

6. Inclusion of Salinas Pond Water Return Facilities as New Source Water Facilities requires execution of a separate agreement between the Parties.

MCWRA and M1W are cooperatively working through the conditions in Section 16.15 but they have not all been completed. The construction of the Blanco Drain and Reclamation Ditch New Source Water Facilities was completed, and they are operational. The capital cost of the New Source Water Facilities is funded by M1W through grants and a low-interest loan from the State of California, Water Resources Control Board, State Revolving Funds with the first payment due on December 31, 2020.

An Amendment to extend the MCWRA payment schedule until June 30, 2020 was granted in June of 2019 through Amendment No. 1 of the Restated Agreement. An Amendment to extend the MCWRA payment schedule until June 30, 2021 was granted in June of 2020 through Amendment No. 2 of the Restated Agreement. Additional time is needed for the conditions in Section 16.15 to be reasonably satisfied which is estimated to be one year. During this time, M1W has the right to full use of the Blanco Drain and Reclamation Ditch New Source Waters until such time the conditions in Section 16.15 have been satisfied. All other terms and conditions of the Agreement remain unchanged and in full force.

OTHER AGENCY INVOLVEMENT:

Monterey One Water participated in the preparation of Amendment No. 3.

FINANCING:

This proposed Amendment No. 3 does not have a financial impact. The Restated Agreement's Conditions Precedent include developing an Engineer's Report and a successful assessment process for rates and charges related to the New Source Water Facilities.

Prepared by: Shaunna Murray, Senior Water Resources Engineer, (831) 755-4860

pproved by: Brent Buche, General Manager, (831) 755-4860

Attachments:

- 1. Amendment No. 3
- 2. Amendment No. 2
- 2. Amendment No. 1
- 3. Amended and Restated Water Recycling Agreement





Board Report

Legistar File Number: WRAPA 21-026

June 04, 2021

Item No.

Board of Supervisors Chambers

168 W. Alisal St., 1st Floor Salinas, CA 93901

Introduced: 5/28/2021

Version: 1

Current Status: Agenda Ready Matter Type: WRA P&A Item

Consider supporting a recommendation that the Monterey County Water Resources Agency Board of Supervisors:

a. Approve Amendment No. 3 to the Amended and Restated Water Recycling Agreement with Monterey One Water for a time extension related to the Conditions Precedent for the New Source Water Facilities; and

b. Request the Chair of the Monterey County Water Resources Agency Board of Supervisors to execute the Amendment.

RECOMMENDATION:

It is recommended that the Personnel and Administration Committee support a recommendation by the Monterey County Water Resources Agency that the Monterey County Water Resources Agency Board of Supervisors:

a. Approve Amendment No. 3 to the Amended and Restated Water Recycling Agreement with Monterey One Water for a time extension related to the Conditions Precedent for the New Source Water Facilities; and

b. Request the Chair of the Monterey County Water Resources Agency Board of Supervisors to execute the Amendment.

SUMMARY/DISCUSSION:

The Monterey County Water Resources Agency (MCWRA) and Monterey One Water (M1W), formerly known as Monterey Regional Water Pollution Control Agency (MRWPCA), entered into an Amended and Restated Water Recycling Agreement (Restated Agreement), which included consideration of the financing, design, construction, operation, maintenance, and replacement of New Source Water Facilities to provide additional recycled water to MCWRA for use in the existing Castroville Seawater Intrusion Project (CSIP), a coastal irrigation project. In addition, M1W would be provided approximately 4,320 AFY of new source water to supplement the Pure Water Monterey Groundwater Replenishment Project which has been developed to provide drinking water.

Portions of the Amended and Restated Water Recycling Agreement applicable to the New Source Water Facilities will not become effective until the following conditions in Section 16.15 are all met:

1. Water Rights for the Blanco Drain and Reclamation Ditch are obtained from the California State Water Resources Control Board.

2. A fully executed, and California Public Utilities Commission approved, Water Purchase Agreement, between MRWPCA, MPWMD, and California-American Water, is approved by the California Public Utilities Commission and executed by the parties thereto. 3. Written finding by the Regional Water Quality Control Board that utilization of the Blanco Drain dry weather flows as New Source Water meets all treatment requirements for the aforesaid dry weather flows.

4. An independent third-party review of proposed capital and operating costs and preparation of an Engineer's Report is approved by the MCWRA Board of Directors and Board of Supervisors. The costs of the aforesaid third-party review shall be shared equally between MCWRA and MRWPCA.

5. A successful assessment or Proposition 218 process for rates and charges related to the operation and maintenance of the New Source Water Facilities and proportional primary and secondary treatment charges.

6. Inclusion of Salinas Pond Water Return Facilities as New Source Water Facilities requires execution of a separate agreement between the Parties.

MCWRA and M1W are cooperatively working through the conditions in Section 16.15 but they have not all been completed. The construction of the Blanco Drain and Reclamation Ditch New Source Water Facilities was completed, and they are operational. The capital cost of the New Source Water Facilities is funded by M1W through grants and a low-interest loan from the State of California, Water Resources Control Board, State Revolving Funds with the first payment due on December 31, 2020.

An Amendment to extend the MCWRA payment schedule until June 30, 2020 was granted in June of 2019 through Amendment No. 1 of the Restated Agreement. An Amendment to extend the MCWRA payment schedule until June 30, 2021 was granted in June of 2020 through Amendment No. 2 of the Restated Agreement. Additional time is needed for the conditions in Section 16.15 to be reasonably satisfied which is estimated to be one year. During this time, M1W has the right to full use of the Blanco Drain and Reclamation Ditch New Source Waters until such time the conditions in Section 16.15 have been satisfied. All other terms and conditions of the Agreement remain unchanged and in full force.

OTHER AGENCY INVOLVEMENT:

Monterey One Water participated in the preparation of Amendment No. 3.

FINANCING:

This proposed Amendment No. 3 does not have a financial impact. The Restated Agreement's Conditions Precedent include developing an Engineer's Report and a successful assessment process for rates and charges related to the New Source Water Facilities.

Prepared by: Shaunna Murray, Senior Water Resources Engineer, (831) 755-4860

pproved by: Brent Buche, General Manager, (831) 755-4860

Attachments:

- 1. Amendment No. 3
- 2. Amendment No. 2
- 2. Amendment No. 1
- 3. Amended and Restated Water Recycling Agreement

AMENDMENT NO. 3 TO AMENDED AND RESTATED WATER RECYCLING AGREEMENT BETWEEN MONTEREY COUNTY WATER RESOURCES AGENCY AND MONTEREY ONE WATER

THIS AMENDMENT NO. 3 to the Amended and Restated Water Recycling Agreement, dated November 3, 2015 (hereinafter, "Agreement") between the Monterey County Water Resources Agency, a political subdivision of the State of California (hereinafter, "WRA") and Monterey One Water (hereinafter, "M1W", referred to in the Agreement as "PCA") is hereby entered into between WRA and M1W (collectively, WRA and M1W are referred to as the "Parties").

WHEREAS, the portions of Agreement applicable to the New Source Water Facilities will not become effective until the following conditions in Section 16.15 are met:

- 1. Water Rights for the Blanco Drain and Reclamation Ditch are obtained from the California State Water Resources Control Board; and
- 2. A fully executed, and California Public Utilities Commission approved, Water Purchase Agreement, between MRWPCA, MPWMD, and California-American Water, is approved by the California Public Utilities Commission and executed by the parties thereto; and
- 3. Written finding by the Regional Water Quality Control Board that utilization of the Blanco Drain dry weather flows as New Source Water meets all treatment requirements for the aforesaid dry weather flows; and
- 4. An independent third-party review of proposed capital and operating costs and preparation of an Engineer's Report is approved by the MCWRA Board of Directors and Board of Supervisors. The costs of the aforesaid third-party review shall be shared equally between MCWRA and MRWPCA; and
- 5. A successful assessment or Proposition 218 process for rates and charges related to the operation and maintenance of the New Source Water Facilities and proportional primary and secondary treatment charges; and
- 6. Inclusion of Salinas Pond Water Return Facilities as New Source Water Facilities requires execution of a separate agreement between the Parties.

WHEREAS, the status of the conditions in Section 16.15 are as follows; conditions 1 and 2 are satisfied; conditions 3,4, and 5 are pending; and condition 6 has not yet commenced.

WHEREAS, the capital cost of the New Source Water Facilities are funded by M1W through grants and a low-interest loan from the State of California, Water Resources Control Board, State Revolving Funds with the first payment made in December 2020.

WHEREAS, if all conditions in Section 16.15 are satisfied, the WRA's share of the capital costs and the repair and replacement costs associated with the New Source Water Facilities are 45.1%.

WHEREAS, if all conditions in Section 16.15 are satisfied, the WRA shall pay M1W the proportional share of the capital cost loan payment 30 days prior to the due date and the proportional share of the amortized replacement/renewal costs by March 1st, annually.

WHEREAS, the construction of the New Source Water Facilities on the Blanco Drain and Reclamation Ditch are complete.

WHEREAS, the Parties entered into Amendment No. 1 to the Agreement on July 18, 2019 to extend the term of satisfying conditions described in Section 16.15 to June 30, 2020 with no associated dollar amount increase to continue to provide services identified in the Agreement.

WHEREAS, the Parties entered into Amendment No. 2 to the Agreement on July 14, 2020 to extend the term of satisfying conditions described in Section 16.15 to June 30, 2021 with no associated dollar amount increase to continue to provide services identified in the Agreement.

WHEREAS, the Parties now wish to amend the Agreement to extend the term of satisfying conditions described in Section 16.15 to June 30, 2022 with no associated dollar amount increase to continue to provide services identified in the Agreement.

NOW, THEREFORE, the Parties agree to amend the Agreement as follows:

1. Amend Section 7.03, "Payment Schedule", to read as follows:

> Modify section 1(c) to have first payment due in 2022, and modify section 1(f) to have first payment due in 2023.

- 2. M1W has the right to utilize the New Source Waters from the Blanco Drain (Water Right Application No. 32263A) and Reclamation Ditch (Water Right Application No. 32263B) Diversion Facilities in full until such time the conditions set forth in Section 16.15 have been satisfied, and in connection therewith, and until such a time the provisions of Section 16.16 shall not be in operation or effect.
- 3. All other terms and conditions of the Agreement remain unchanged and in full force.
- 4. This Amendment No. 3 shall be attached to the Agreement as amended and incorporated therein as if fully set forth in the Agreement.

IN WITNESS WHEREOF, the duly authorized representatives of the Parties hereto have executed this Amendment No. 3 to the Agreement as of the day and year first written below:

MONTEREY COUNTY WATER **RESOURCES AGENCY**

MONTEREY ONE WATER

By:	
-----	--

General Manager

By:___

General Manager

Date:

Date: _____

Approved as to Form

By:	By:
Deputy County Counsel	Counsel, Monterey One Water
Date:	Date:
Approved as to Fiscal Provisions	
By:Auditor-Controller	
Date:	
Approved as to Indemnity, Insurance Provisions	
By:	
Risk Management	
Date:	

AMENDMENT NO. 2 TO AMENDED AND RESTATED WATER RECYCLING AGREEMENT BETWEEN MONTEREY COUNTY WATER RESOURCES AGENCY AND MONTEREY ONE WATER

THIS AMENDMENT NO. 2 to the Amended and Restated Water Recycling Agreement, dated November 3, 2015 (hereinafter, "Agreement") between the Monterey County Water Resources Agency, a political subdivision of the State of California (hereinafter, "WRA") and Monterey One Water (hereinafter, "M1W", referred to in the Agreement as "PCA") is hereby entered into between WRA and M1W (collectively, WRA and M1W are referred to as the "Parties").

WHEREAS, the portions of Agreement applicable to the New Source Water Facilities will not become effective until the following conditions in Section 16.15 are met:

- 1. Water Rights for the Blanco Drain and Reclamation Ditch are obtained from the California State Water Resources Control Board; and,
- 2. A fully executed, and California Public Utilities Commission approved, Water Purchase Agreement, between MRWPCA, MPWMD, and California-American Water, is approved by the California Public Utilities Commission and executed by the parties thereto; and,
- 3. Written finding by the Regional Water Quality Control Board that utilization of the Blanco Drain dry weather flows as New Source Water meets all treatment requirements for the aforesaid dry weather flows; and,
- 4. An independent third-party review of proposed capital and operating costs and preparation of an Engineer's Report is approved by the MCWRA Board of Directors and Board of Supervisors. The costs of the aforesaid third-party review shall be shared equally between MCWRA and MRWPCA; and,
- 5. A successful assessment or Proposition 218 process for rates and charges related to the operation and maintenance of the New Source Water Facilities and proportional primary and secondary treatment charges; and,
- 6. Inclusion of Salinas Pond Water Return Facilities as New Source Water Facilities requires execution of a separate agreement between the Parties; and

WHEREAS, the status of the conditions in Section 16.15 are as follows; conditions 1 and 2 are satisfied; conditions 3,4, and 5 are pending; and condition 6 has not yet commenced; and

WHEREAS, the capital cost of the New Source Water Facilities are funded by M1W through grants and a low-interest loan from the State of California, Water Resources Control Board, State Revolving Funds with the first payment due in December 2020 and

WHEREAS, if all conditions in Section 16.15 are satisfied, the WRA's share of the capital costs and the repair and replacement costs associated with the New Source Water Facilities are 45.1%; and

WHEREAS, if all conditions in Section 16.15 are satisfied, the WRA shall pay M1W the proportional share of the capital cost loan payment 30 days prior to the due date and the proportional share of the amortized replacement/renewal costs by March 1st, annually; and

WHEREAS, the construction of the New Source Water Facilities on the Blanco Drain and Reclamation Ditch are complete; and

WHEREAS, the Parties amended the Agreement on July 18, 2019 to extend the term of satisfying conditions described in Section 16.15 to June 30, 2020 with no associated dollar amount increase to continue to provide services identified in the Agreement; and

WHEREAS, the Parties now wish to amend the Agreement to extend the term of satisfying conditions described in Section 16.15 to June 30, 2021 with no associated dollar amount increase to continue to provide services identified in the Agreement.

NOW, THEREFORE, the Parties agree to amend the Agreement as follows:

1. Amend Section 7.03, "Payment Schedule", to read as follows:

Modify sections 1(c) to have first payment due in 2021, and modify section 1(f) to have first payment due in 2022.

- 2. MIW has the right to utilize the New Source Waters from the Blanco Drain (Water Right Application No. 32263A) and Reclamation Ditch (Water Right Application No. 32263B) Diversion Facilities in full until such time the conditions set forth in Section 16.15 have been satisfied, and in connection therewith, and until such a time the provisions of Section 16.16 shall not be in operation or effect.
- 3. All other terms and conditions of the Agreement remain unchanged and in full force.
- 4. This Amendment No. 2 shall be attached to the Agreement as amended and incorporated therein as if fully set forth in the Agreement.

IN WITNESS WHEREOF, the duly authorized representatives of the Parties hereto have executed this Amendment No. 2 to the Agreement as of the day and year first written below:

MONTEREY COUNTY WATER RESOURCES AGENCY

General Manager

7/14/2020

Date:

By:

Approved as to Form

By: Kelly L. Donlon Deputy County Counsel

Date: 7/14/20

Page 2 of 3

MONTEREY ONE WATER

General Manager 23 2020 Date:

Bv: Counsel. Monterey One Water

7/8/2020 Date:

Amendment No. 2 Monterey One Water

J

Approved as to Fiscal Provisions		
Jary K Giboney		
By:		
Auditor-Controller		

Date: 7/14/2020

Approved as to Indemnity, Insurance Provisions

By: ______Risk Management

Date: _____

AMENDMENT NO. 1 TO AMENDED AND RESTATED WATER RECYCLING AGREEMENT BETWEEN MONTEREY COUNTY WATER RESOURCES AGENCY AND MONTEREY ONE WATER

THIS AMENDMENT NO. 1 to the Amended and Restated Water Recycling Agreement, dated November 3, 2015, (hereinafter, "Agreement") between the Monterey County Water Resources Agency, a political subdivision of the State of California (hereinafter, "WRA") and Monterey One Water (hereinafter, "M1W," referred to in the Agreement as "PCA") is hereby entered into between WRA and M1W (collectively, WRA and M1W are referred to as the "Parties").

WHEREAS, the portions of Agreement applicable to the New Source Water Facilities will not become effective until the following conditions in Section 16.15 are met:

- 1. Water Rights for the Blanco Drain and Reclamation Ditch are obtained from the California State Water Resources Control Board; and,
- 2. A fully executed, and California Public Utilities Commission approved, Water Purchase Agreement, between MRWPCA, MPWMD, and California-American Water, is approved by the California Public Utilities Commission and executed by the parties thereto; and,
- 3. Written findings are made by the Regional Water Quality Control Board that utilization of
- the Blanco Drain dry weather flows as New Source Water meets all treatment requirements for the aforesaid dry weather flows; and,
- 4. An independent third-party review of proposed capital and operating costs and preparation of an Engineer's Report is approved by the WRA Board of Directors and Board of Supervisors. The costs of the aforesaid third-party review shall be shared equally between WRA and M1W; and,
- 5. A successful assessment or Proposition 218 process for rates and charges related to the operation and maintenance of the New Source Water Facilities and proportional primary and secondary treatment charges; and,
- 6. Inclusion of Salinas Pond Water Return Facilities as New Source Water Facilities requires execution of a separate agreement between the Parties; and,

WHEREAS, the status of the conditions in Section 16.15 are as follows; conditions 1 and 2 are satisfied; conditions 3,4, and 5 are pending; and condition 6 has not yet commenced; and,

WHEREAS, the capital cost of the New Source Water Facilities are funded by M1W through grants and a low-interest loan from the State of California, Water Resources Control Board, State Revolving Funds with the first payment due on December 31, 2019; and,

WHEREAS, if all conditions in Section 16.15 are satisfied, the WRA's share of the capital costs and the repair and replacement costs associated with the New Source Water Facilities are 45.1%; and, 2004 15 1 1 2 1 1 2

1. 2. 1

WHEREAS, if all conditions in Section 16.15 are satisfied, the WRA shall pay M1W the proportional share of the capital cost loan payment annually 30 days prior to the due date and the proportional share of the amortized replacement/renewal costs annually by March 1st; and,

WHEREAS, the construction of the New Source Water Facilities will be substantially complete on August 23, 2019; and,

WHEREAS, the Parties now wish to amend the Agreement to extend the term of satisfying conditions described in Section 16.15 to June 30, 2020 with no associated dollar amount increase to continue to provide services identified in the Agreement.

NOW, THEREFORE, the Parties agree to amend the Agreement as follows:

1. Amend Section 7.03, "Payment Schedule", to read as follows:

Modify section 1(c) to have the first payment due in 2020, and modify section 1(f) to have first payment due in 2021.

- 2. M1W has the right to utilize the New Source Water in full until such time the conditions set forth in Section 16.15 have been satisfied, and in connection therewith, and until such time, the provisions of Section 16.16 shall not be in operation or effect.
- 3. All other terms and conditions of the Agreement remain unchanged and in full force.
- 4. This Amendment No. 1 shall be attached to the Agreement and incorporated therein as if fully set forth in the Agreement.

IN WITNESS WHEREOF, the duly authorized representatives of the Parties hereto have executed this Amendment No. 1 to the Agreement as of the day and year first written below:

MONTEREY COUNTY WATER **RESOURCES AGENCY** By: General Manager 8 Date:

Approved as to Form By: Deputy County Counsel Date:

MONTEREY ONE WATER

By: General Manager

(122/1 Date:

By: Monterey One Water 7/12/19 Date:

Approved as	to Fiscal Provisions
Ву:	Auditor-Controller
Date:	8-21-19

Approved as to Indemnity, Insurance Provisions

By: <u>n/q</u> K-CV Risk Management

Date: _____

.

•

Amended and Restated Water Recycling Agreement Between Monterey Regional Water Pollution Control Agency and Monterey County Water Resources Agency





Table of Contents

I. CON	NSTRUCTION OF THE NEW SOURCE WATER FACILITIES	9
1.01.	PCA to construct New Source Water Facilities.	9
1.02.	Change orders	9
1.03.	Location of the New Source Water Facilities.	9
1. 04.	Projected Future New Source Water Facilities.	9
1.05.	Cooperation between agencies	. 10
1. 06.	Regular meetings	. 10
II. EST	'IMATED COSTS AND FINANCING FOR NEW SOURCE WATER FACILITIES	.10
2.01.	New Source Water Facilities.	. 10
2.02.	Estimated design and construction costs of New Source Water Facilities.	. 11
2.03.	Source Water	.11
	NERSHIP, OPERATION AND MAINTENANCE OF CSIP, SVRP, SRDF AND NE	
	JRCE WATER FACILITIES.	
3.01.	Ownership, operation, and maintenance, in general.	
3.02,	Safety and loss prevention program	
3.03.	Warranties	
3.04.	Right to inspect.	
3.05.	Daily operation.	
3.06.	Incidental uses	
3.07.	Notice of temporary cessation of water deliveries	
3.08,	Outside Contracts	
3.09.	Access to CSIP and SRDF facilities.	
IV. PRC	OVISION OF RECYCLED WATER TO WRA FROM PCA	
4.01.	Existing Allocations	
4.02.	New Incremental and Interruptible Allocations (New Source Water) - Phase I	
4.03.	Operations and Maintenance Treatment Cost Allocation	. 15
4.04.	New Incremental and Interruptible Allocations – Phase II.	. 15
4.05.	Provisions for Drought Reserve.	
4.06.	Water Quality Enhancement Options.	
V. DEI	LIVERY OF WATER TO WRA	
5.01.	Quantity of water to be delivered to WRA SVRP.	. 17
5.02.	WRA's demand schedule	. 17
5.03.	WRA's demand.	17
5.04.	PCA'S duty to comply with WRA demand.	17

VI. MAI	NTENANCE OF WATER QUALITY	17
6.01.	Water Quality	17
6.02.	Duty to monitor water quality; cessation in deliveries.	18
6.03.	Regulations to protect water quality.	18
6.04.	Water Quality and Operations Committee	18
VII. PAY	MENTS BY WRA TO PCA	19
7.01.	Consideration paid by WRA	19
7.02.	Amounts to be paid	19
7.03.	Payment schedule.	19
7.04.	Payment procedures	20
7.05.	Reduction or termination of water delivery for lack of payment; judicial remedies	20
7.06.	Prior approval of budget by WRA.	21
7.07.	SWRCB as third party beneficiary.	21
7.08.	Priorities of payment	21
7.09.	Application of loan payments by PCA.	22
7.10.	Unanticipated events/Emergency situations	22
7.11.	Decision-making authority	22
VIII. ACC	COUNTING SYSTEM, REPORTS	23
8.01.	Accounting system	23
8.02.	Financial reports	23
8.03.	Direct and indirect costs	23
8.04.	Annual audit	24
8.05.	Right to inspect and audit records	24
8.06.	Reimbursement for overcharge or undercharge	24
IX. REPA	AIR AND MODIFICATION OF SVRP, CSIP, SRDF, AND NEW SOURCE WA' ILITIES AND RELATED FACILITIES	ГER 24
9.01.	SVRP, CSIP, and SRDF Reserve Funds	24
9.02.	Replacement and Renewal Costs Associated with New Source Water Facilities	25
9.03.	Expansion of treatment plant capacity	25
9.04.	Interruptions of service.	25
9.05.	Duty to apply insurance proceeds.	25
9.06.	Payment of uninsured losses	25
9.07.	New Water Recycling Agreement on replacement of SVRP.	26
X. INDE	EMNIFICATION	26
10.01.	Indemnification of WRA.	
10.02.	Indemnification of PCA	26

10.03.	Procedure for Indemnification	
10.04.	Payment of indemnified claims.	
10.05.	Contribution in the event of shared liability.	
10.06.	Exclusion from O&M costs	
XI. INSU	URANCE	
11.01.	General insurance requirements	
11.02.	Commercial general liability insurance.	
11. 03.	Motor vehicle insurance	
11 .04 .	Property insurance.	
11.05.	Workers' compensation insurance	
11.06.	Certificate of insurance	
11. 07.	Self-insurance.	
11 .08 .	Insurance costs	
11 .09 .	Periodic increases in coverage requirements.	
XII. TER	M OF AGREEMENT	
12.01.	Term of Agreement	
12.02.	Automatic extension	
12.03.	Conditions of agreement during term.	
12.04.	Rights on Termination.	
XIII. DIS	PUTE RESOLUTION.	
13.01.	Dispute resolution procedure.	
13.02.	Duty to meet and confer	
13.03.	Mediation and Arbitration	
XIV. ADI	DITIONAL RESPONSIBILITIES	
14.01.	Compliance with water reclamation requirements for the CSIP	
14.02.	Grower and landowner requirements	
14.03.	Reports to U.S. Department of the Interior, Bureau of Reclamation	
14.04.	Permits and approvals	
14.05.	Third-party damages	
1 4.06 .	Assessments.	
14.07.	Physical damage to crops	
XV. DEL	IVERY OF WATER TO PCA.	
15.01.	PCA Demand	
15.02.	PCA Monthly Demand Schedule	
15.03.	PCA Payment	

15.04.	No Modification of MCWD Contract Entitlement
XVI. GEN	ERAL PROVISIONS
16.01.	Compliance with laws
16.02.	Attorney's fees
16.03.	Amendments
16.04.	Contract administrators
16.05.	Assignment
16.06.	Negotiated Water Recycling Agreement
16.07.	Time is of essence
16.08.	Headings
16.09.	Entire Water Recycling Agreement
16.10.	Notices
16.11.	Execution of documents
16.12.	Exhibits
16.13.	Severability
16.14.	Waiver
16.15.	Conditions Precedent for New Source Water Facilities
16.16.	Use of New Source Water
16.17.	Written Authorization
	ECUTION
EXHIBI	
DVIIID	SHOWING POINTS OF DELIVERY TO THE CSIP
EXHIBI	T B: OPERATION AND MAINTENANCE OF SVRP, CSIP, SRDF, AND NEW SOURCE WATER FACILITIES
EXHIBI	
EXHIBI	D: WASTEWATER RECLAMATION SYSTEM PROJECT
	DESCRIPTION AND MITIGATION MEASURES
EXHIBI	
EXHIBI	OF THIS WATER RECYCLING AGREEMENT F: OPERATION AND MAINTENANCE ESTIMATED TREATMENT COSTS
EXHIBI	
EXHIBI	
EXHIBI	
	NEW SOURCE WATER FACILITIES
EXHIBI	J: SAMPLE COST ALLOCATION PLAN FOR INDIRECT COSTS

The PCA was formed as a California Joint Powers Agency by a Joint Exercise of Powers Agreement for the Monterey Regional Water Pollution Control Agency, effective June 29, 1979. The Monterey County Water Resources Agency ("WRA") is a flood control and water agency established by the legislature in 1990 (Monterey County Water Resources Agency Act, Water Code Appendix Chapter 52), and is the successor to the Monterey County Flood Control and Water Conservation District.

RECITALS

WHEREAS, PCA and WRA entered into the Memorandum of Understanding Regarding Source Waters and Water Recycling, as amended, dated October 8, 2014, with the City of Salinas ("Salinas"), Marina Coast Water District ("MCWD"), and Monterey Peninsula Water Management District ("WMD"), hereinafter referred to as "Five Party MOU," to provide a framework for provision of additional source waters that could be dedicated to the Pure Water Monterey Project and provide additional water supply to the Castroville Seawater Intrusion Project; and,

WHEREAS, PCA and WRA had previously entered into an agreement, dated June 16, 1992, for construction and operation of a tertiary treatment system (the "1992 Agreement"), with subsequent amendments thereto, as follows: Amendment No. 1 on May 30, 1994; Amendment No. 2 on February 16, 1998; and Amendment No. 3 on May 28, 2002; and,

WHEREAS, PCA and WRA entered into an agreement between the Monterey County Water Resources Agency and The Monterey Regional Water Pollution Control Agency for the Operation and Maintenance of the Salinas River Diversion Facility, dated February 3, 2011 ("SRDF Agreement"); and,

WHEREAS, in order to facilitate coordination of and accounting for operation and maintenance of the Salinas Valley Reclamation Project ("SVRP"), the Castroville Seawater Intrusion Project ("CSIP") and the Salinas River Diversion Facility ("SRDF"), it is desired by the Parties to have one comprehensive agreement; and,

WHEREAS, the Parties desire to facilitate acquisition of new source waters required for the Pure Water Monterey Project and additional waters for CSIP, WRA applied to the California State Water Quality Control Board on May 19, 2014 for water rights to 25,000 acre-ft/year of Drainage Flows from the Blanco Drain, the Reclamation Ditch, and the Tembladero Slough, as referenced in Section 1.0 D herein; and,

WHEREAS, on July 29, 2015 WRA requested that its water rights application be divided into five separate applications with the first 6,500 acre-ft/year allocated to Application Nos. 32263A, 32263B, and 32263C as part of New Incremental and Interruptible Allocations – Phase I, associated with the Pure Water Monterey Project; and,

WHEREAS, PCA and Salinas will enter into an agreement for primary and secondary treatment by PCA of the City's industrial pond water; and,

WHEREAS, the 1992 Agreement, as amended, needs to be replaced in order to fulfill the terms and conditions of the Five-Party MOU, and to update specific terms and conditions of agreement between the Parties, for current needs and realities, while preserving the original intentions of the Parties to the maximum extent possible; and,

WHEREAS, this Agreement relates to and implements certain portions of the Pure Water Monterey Groundwater Replenishment Project ("GWR") that the MRWPCA Board approved on October 8, 2015. The MRWPCA Board certified the Environmental Impact Report ("EIR") for the GWR Project as complete and in compliance with the California Environmental Quality Act ("CEQA"), and adopted the findings required by CEQA on October 8, 2015. This Agreement does not change the GWR Project and no change of circumstances or new information herein shows that the GWR Project would result in new or substantially more severe environmental impacts such that major revisions to the certified EIR would be required. This Agreement is approved based on the EIR as certified.

NOW, THEREFORE, for and in reliance of the foregoing, the parties hereby agree as follows:

1.0 <u>Definitions</u>

For the purposes of this Water Recycling Agreement, the following definitions are provided:

A. The term "Annexation Agreements" refers to the Annexation Agreement between MCWD and PCA dated April 25, 1989, and the Annexation Agreement between MCWD and WRA dated March 26, 1996. The individual Annexation Agreements are referenced herein by their respective dates.

B. The term "Bureau Loan Contract" refers to the Contract Between the United States and the Monterey Regional Water Pollution Control Agency (MRWPCA) for "A Loan for Construction of a Small Reclamation Plant," dated June 2, 1995,

C. The terms "Castroville Irrigation System," "Castroville Seawater Intrusion Project," and "CSIP," as used in this Water Recycling Agreement, refer to the tertiary treated water distribution system and are hereinafter collectively referred to simply as the "CSIP", which is owned by WRA. Reclaimed water produced by the SVRP for CSIP will be delivered to a distribution system, known as the Castroville Irrigation System, at the points indicated on Exhibit A, attached hereto and made a part hereof, as may be amended by agreement of the parties, and through that system it will be delivered to growers in the Castroville area, for use in the irrigation of crops.

D. The term "Drainage Flows" refers to WRA's portion of New Source Waters originating from Blanco Drain, Reclamation Ditch, and Tembladero Slough.

E. The term "Drought Reserve" shall refer to storage of up to 1,000 acre-ft of water for potential use during a drought.

F. The term "Interruptible Rate" applies to PCA charges for primary and secondary treatment of New Source Waters.

G. The term "New Source Water Facilities" applies to facilities required to be constructed to convey Blanco Drain, Reclamation Ditch and Tembladero Slough waters to the PCA Regional Treatment Plant, and those to be constructed to allow wintertime operation of the SVRP.

H. "New Source Waters" are defined as waters originating at the following:

- 1. Agricultural Wash Water
- 2. Blanco Drain Water
- 3. Reclamation Ditch/Tembladero Slough water
- 4. Monterey Storm Water
- 5. Salinas Storm Water

I. The term "Pure Water Monterey Facilities" refers to those advanced treatment facilities necessary to transform secondary treated wastewater from the Regional Treatment Facility into drinking water quality water, the agreement terms for which are not included in this Water Recycling Agreement.

J. The terms "reclaimed water", "reclaimed wastewater", "recycled water", and "tertiary treated water" all refer to the water produced by the 29.6 MGD (33,154 acre-ft per year) tertiary treatment system and are hereinafter collectively referred to as "tertiary treated water".

K. The term "Regional Treatment Plant" refers to the facilities constructed and sized to
provide primary and secondary treatment for 29.6 mgd of wastewater flow and which is owned, operated and maintained by PCA.

L. The terms "SVRP" and "Salinas Valley Reclamation Plant" and "recycled water plant" refer to that part of the Monterey County Water Recycling Project containing the 29.6 mgd (33,154 acre-feet per year) tertiary treatment system that produces recycled water pursuant to Title 22 of the California Code of Regulations (CCR) suitable for irrigation of crops eaten raw, hereinafter collectively referred to simply as the "SVRP," which is owned, operated and maintained by PCA.

M. The terms "SRDF" and "the Salinas River Diversion Facility" both refer to the Salinas River Diversion Facility, which includes the chlorination system as a component, and are hereinafter referred to simply as the "SRDF", which is owned by WRA.

N. The term "SWRCB Loan Contract" is used to refer to "Contract between the State Water Resources Control Board (SWRCB) and MRWPCA for a State Revolving Fund Loan for Construction of a Tertiary Treatment Facility Project" dated December 15, 1995.

O. Member entities of PCA are the following:

- 1. City of Salinas
- 2. City of Monterey
- 3. Castroville Community Services District
- 4. City of Seaside
- 5. City of Pacific Grove
- 6. Marina Coast Water District
- 7. County of Monterey
- 8. City of Del Rey Oaks
- 9. Sand City
- 10. Boronda County Sanitation District

I. CONSTRUCTION OF THE NEW SOURCE WATER FACILITIES

1.01. PCA to construct New Source Water Facilities.

Upon the receipt of final commitments for the financing described below by all applicable governmental agencies and when conditions precedent of Section 16.15 are met, PCA will finance, design, construct, and install the New Source Water Facilities, in substantial conformity with designs and plans approved by the parties in writing.

1.02. Change orders.

Change orders must be approved in writing. Any change order or related set of change orders that increases the New Source Water Facilities cost by \$100,000 or more shall require the written consent of both Parties within 30 days of presentation. Any change order or related set of change orders that increases the New Source Water Facilities cost by less than \$100,000 or that lowers the New Source Water Facilities cost by PCA alone, without the consent of the WRA, except that a copy of any proposed or executed change order shall promptly be provided to WRA as soon as it is available to PCA. Each Party's contract administrator shall be authorized to give consent to change orders for that Party. Neither Party's consent to a change order will be unreasonably withheld or delayed.

1.03. Location of the New Source Water Facilities.

The New Source Water Facilities shall be located as shown in Exhibit G, attached hereto and made a part hereof. PCA will acquire any rights-of-way necessary for the construction and maintenance of pipelines from the sources points of delivery to the Regional Treatment Plant.

1.04. Projected Future New Source Water Facilities.

New Source Water Facilities consist of, but are not limited to the following:

- 1. Blanco Drain Water
 - (a) 2,738 gpm (6.1 cfs) pump station
 - (b) 7,700 feet of 18 inch diameter HDPE, PVC, or Ductile Iron pipe
 - (c) Intake Screen/Wet Well with appropriate appurtenances
- 2. Reclamation Ditch Water
 - (a) 2,693 gpm (6.0 cfs) pump station
 - (b) 43 feet of 6 and 12 inch diameter DIP or PVC discharge pipe
- 3. Tembladero Slough Water
 - (a) 1,346 gpm (3 cfs) pump station
 - (b) 120 feet of 16 inch diameter DIP and PVC discharge pipe
 - (c) Intake screen/Wet Well with appropriate appurtenances
- 4. Monterey Storm Water (Lake El Estero)
 - (a) 2,500+/- gpm pump station
 - (b) 45 feet if 12+/- inch PVC pipe
- 5. Salinas Storm Water
 - (a) Diversion Point No. 1
 - i. 54" x 54" concrete diversion structure
 - ii. 24" x 24" concrete diversion structure plus weir
 - iii. Parshall Flume
 - iv. Intake Screen/Sluice Gate with appropriate appurtenances
 - (b) Diversion Point No. 2
 - i. 66"x 66" concrete diversion structure

- ii. 24"x 24" concrete diversion structure plus weir
- iii. Rehabilitate or new 150 feet of -inch pipe
- iv. Parshall Flume
- v. Appropriate access points of appurtenances
- (c) Diversion Point No. 3
 - i. 30" x 30" concrete diversion structure
 - ii. Parshall Flume with 12"x 6'x 40 feet diversion structure
 - iii. Appropriate access points and appurtenances
- (d) Diversion Point No. 4
 - i. Cast in place diversion structure
 - ii. 80 feet of 30 inch pipe
 - iii. Parshall Flume and appropriate access points and appurtenances
- (e) Salinas Pond Water Return Facilities
 - i. Intake Structure
 - ii. 2 45 hp pumps
 - iii. Valve and Meter boxes
 - iv. 20-inch HDPE 6200 ft
 - v. Appropriate access points and appurtenances
- (f) Salinas Pond Return Facilities
 - i. 10'x16' wet well
 - ii. 2 45 hp pumps
 - iii. 7,500 feet of HDPE pipe
 - iv. Appropriate access points and appurtenances
- 6. Modifications to Regional Treatment Plant
 - (a) Two 30-inch sluice gates in Chlorine Contact Basins
 - (b) One 72-inch motorized sluice gate at Storage Pond with access structure and stairs
 - (c) 860-feet of 30+/- inch HDPE, DIP, PVC, or concrete pipe

1.05. Cooperation between agencies.

PCA and WRA shall work cooperatively and with diligence to obtain all permits, approvals, and financing to construct the New Source Water Facilities. PCA and WRA shall work cooperatively and with diligence to obtain water rights for the new water sources.

1.06. Regular meetings.

After development of an implementation schedule, representatives of the Parties will meet on a monthly basis, or more often if necessary, in order to ensure that the construction of the New Source Water Facilities is proceeding according to the schedule and in conformity with this agreement and the approved plans and designs.

II. ESTIMATED COSTS AND FINANCING FOR NEW SOURCE WATER FACILITIES.

2.01. New Source Water Facilities.

PCA shall bear all the costs of design and construction of the New Source Water Facilities. PCA intends to finance these costs utilizing grants and low interest loans. WRA will repay its share of these costs as provided in Section VII.

2.02. Estimated design and construction costs of New Source Water Facilities.

Component	WRA	PCA	Monterey/Salinas	TOTAL
Blanco Drain Water	\$2.3 Million	\$2.7 Million	0	\$5.0 Million
Reclamation Ditch	\$0.5 Million	\$0.6 Million	0	\$1.1 Million
Tembladero Slough	\$0.5 Million	\$0.6 Million	0	\$1.1 Million
Monterey Storm Water	0	0	Not Estimated	WIT INMION
Salinas Storm Water	0	0	Not Estimated	
Salinas Pond Water Return Facilities	\$1.3 Million	\$1.5 Million		\$2.8 Million
Modifications to Regional Treatment Facility	\$0.6 Million	\$0.7 Million		\$1.3 Million
TOTAL	\$5.2 Million	\$6.1 Million		\$11.3 Million

Table 1.

WRA's share of capital costs is 45.1% and PCA's share of capital costs is 54.9%. This same allocation of capital costs will be applied to Replacement and Renewal costs associated with the New Source Water Facilities, as identified in Section 9.02.

2.03. Source Water.

- 1. WRA shall obtain water rights to Drainage Flows from the California State Water Resources Control Board. Costs of obtaining and maintaining said water rights shall be reimbursed to WRA on the same proportional basis for water used as a component of the New Source Facilities.
- 2. Water rights for New Source Water within the City of Salinas and City of Monterey, and any future new source waters not identified herein, including without limitation, dry and wet weather storm drainage system flows and/or volumes, are not subject to this Agreement and may be the subject of future agreements.

III. OWNERSHIP, OPERATION AND MAINTENANCE OF CSIP, SVRP, SRDF AND NEW SOURCE WATER FACILITIES.

3.01. Ownership, operation, and maintenance, in general.

- 1. PCA will own, operate, and maintain the SVRP as described in Exhibit B, "Operation and Maintenance of SVRP, CSIP, SRDF, and New Source Water Facilities", attached hereto and made a part hereof, and will keep the SVRP in good condition and repair for the term of this Water Recycling Agreement. WRA shall be required to reimburse PCA for such costs and expenses of ownership, operation, and maintenance of the SVRP as described in Article VII of this Water Recycling Agreement. After expiration of this Water Recycling Agreement and any extension thereof, the SVRP will remain the property of PCA, except that WRA will own the discharge pipeline from the PCA property line near the discharge side of the storage pond to the connection with the CSIP.
- 2. WRA will own the CSIP facilities and PCA will operate and maintain the CSIP facilities as described in Exhibit B, "Operation and Maintenance of SVRP, CSIP, SRDF, and New Source Water Facilities", and will keep the CSIP in good condition and repair for the term of this Water Recycling Agreement. WRA shall be required to reimburse PCA for such costs

and expenses of operation and maintenance of the CSIP as described in Article VII of this Water Recycling Agreement. After expiration of this Water Recycling Agreement, and any extension thereof, the CSIP will remain the property of WRA.

- 3. WRA will own, and PCA will operate and maintain the SRDF as described in Exhibit B, "Operation and Maintenance of SVRP, CSIP, SRDF, and New Source Water Facilities", and will keep the SRDF in good condition and repair for the term of this Water Recycling Agreement. WRA shall be required to reimburse PCA for such costs and expenses of operation and maintenance of the SRDF as described in Article VII of this Water Recycling Agreement. After expiration of this Water Recycling Agreement and any extension thereof, the SRDF will remain the property of WRA.
- 4. PCA will own, operate, and maintain the New Source Water Facilities as described in Exhibit B, "Operation and Maintenance of SVRP, CSIP, SRDF, and New Source Water Facilities", and will keep the New Source Water Facilities in good condition and repair for the term of this Water Recycling Agreement. WRA shall be required to reimburse PCA proportionately for such costs and expenses of ownership, operation, and maintenance of the New Source Water Diversion Facilities as described in Article VII of this Water Recycling Agreement. After expiration of this Water Recycling Agreement and any extension thereof, the New Source Water Diversion Facilities will remain the property of PCA.
- 5. No changes in ownership, operation or maintenance of SVRP, CSIP, SRDF or the New Source Water Facilities shall occur except with the mutual consent of both parties.

3.02. Safety and loss prevention program.

PCA will develop, maintain, and implement a safety and loss prevention program for the SVRP, CSIP, SRDF and New Source Water Facilities, and will provide appropriate training for its employees working on the SVRP. This program will conform with all requirements set forth in CAL OSHA's Process Safety Management Program and US EPA's Risk Management Program, and will be revised and updated as new regulations are promulgated. All costs associated with the program will be included in the annual budget process to be reimbursed by WRA.

3.03. Warranties.

PCA warrants that all water committed to WRA pursuant to this Water Recycling Agreement shall be transferred to WRA free and clear of all claims by any person or entity, and that all water produced by the SVRP, up to the SVRP's capacity of 29.6 MGD, shall be distributed in conformity with Section IV.

3.04. Right to inspect.

WRA shall have the right to inspect the SVRP and New Source Water Facilities, while under construction and at any time thereafter during the term of this contract, upon the giving of reasonable advance notice to PCA. Such inspections may take place at any time during the day or night; however, night time inspections will not take place without at least one week's notice, except in case of emergency or by agreement between the Parties. However, PCA shall have the sole right to direct the construction work and the work of PCA's own employees. WRA's right to inspect is for any lawful purpose, but not for the purpose of supervision of the work observed.

3.05. Daily operation.

The SVRP, the CSIP, SRDF and New Source Water Diversion Facilities will be in operation and will

supply water to WRA on a daily basis when source water is available as described in Exhibit B "Operation and Maintenance of SVRP, CSIP, SRDF, and New Source Water Facilities", except for temporary periods of shut-down authorized by this Water Recycling Agreement or made necessary by circumstances beyond the control of PCA.

3.06. Incidental uses.

PCA may use such amounts of tertiary treated water from the SVRP as may be needed for the normal operation and maintenance of PCA's facilities, including, but not limited to, the SVRP and the primary and secondary treatment facilities.

3.07. Notice of temporary cessation of water deliveries.

PCA will give immediate notice to WRA, by telephone to WRA's General Manager, or to the person designated by the General Manager to receive such notices, with a prompt follow-up notice in writing, as soon as PCA becomes aware of the need to cease deliveries to the CSIP, whatever may be the reason for such interruption in service. Whenever an unforeseen cessation of deliveries occurs without prior notice to WRA, PCA shall immediately give notice to WRA as provided above. In addition, whenever a cessation of deliveries occurs, PCA shall use every reasonable effort to restore service as soon as possible.

3.08. Outside Contracts.

When PCA deems it more appropriate for someone other than PCA to make a repair directly, PCA will obtain contracts to perform this work through bidding or other appropriate competitive procurement process.

3.09. Access to CSIP, SRDF, and New Source Water Facilities on WRA Land or Easements.

WRA shall provide the necessary access arrangements for PCA personnel to carry out their required work on the CSIP, SRDF, and New Source Water Facilities on WRA land or Easements. WRA shall notify landowners, growers, and others who may be affected by this work in advance that PCA personnel will be entering onto and performing work on their property. Any disputes arising between PCA personnel and these affected parties will cause PCA to discontinue work on the affected facilities until WRA has established necessary access arrangements for the work to continue.

IV. PROVISION OF RECYCLED WATER TO WRA FROM PCA

4.01. Existing Allocations

- 1. WRA shall be entitled to tertiary treated recycled water for its CSIP Project during the agricultural growing season in a volume not less than total wastewater flows to the Regional Treatment Plant from all PCA members existing at the Effective Date of this Water Recycling Agreement, plus all other areas within PCA's 2001 boundaries less the following amounts (may be taken before tertiary treatment):
 - (a) Amount claimed and utilized by MCWD pursuant to Section 15.04 as provided pursuant to the Annexation Agreements.
 - (b) Such flows as are lost or as must be diverted in the ordinary course of operating and maintaining the treatment plant and ocean outfall.
 - (c) Such flows as are not needed to meet WRA's authorized demand pursuant to this Water Recycling Agreement.
 - (d) 650 AF of water allocated by WRA to PCA per Table 2:

Table 2	2.
---------	----

Month	Typical Monthly Seasonal Spread (AF)
May	138 AF
June	172 AF
July	185 AF
August	155 AF
TOTAL:	650 AF

- 2. WRA shall be entitled to one-half of the volume of wastewater flows from areas outside of PCA's 2001 Boundary provided; however, at the request of WRA, PCA passes the wastewater flows through the tertiary treatment facility or Pure Water Monterey Facilities.
 - (a) PCA shall be responsible for incremental tertiary treatment operations, maintenance, repair and replacement costs for any and up to one-half of the flows from areas outside PCA's 2001 Boundary that pass through the tertiary treatment facility and that are utilized by PCA.
 - (b) WRA shall be responsible for incremental tertiary treatment operations, maintenance, repair and replacement costs for any and up to one-half of the flows from areas outside PCA's 2001 Boundary that pass through the tertiary treatment facility and that are utilized by WRA.
 - (c) PCA, for its flows pursuant to Section 4.02 (2)(a) above, will comply with all applicable requirements set forth in Contract No. 5-07-20-W1284, between the Bureau of Reclamation and WRA including, but not limited to, those contained in the Bureau Loan Agreement, all at PCA's sole cost and expense.
- 4.02. New Incremental and Interruptible Allocations (New Source Water) Phase I
 - 1. WRA and PCA agree that Phase I consists of PCA constructing and operating New Source Water Facilities intended to provide New Source Water of approximately 4,320 acre-ft/year for PCA and 4,381 acre-ft/year for WRA. Of the total volume of approximately 8,701 acre-ft/year, PCA's operation and maintenance cost share is 49.7% and WRA's operation and maintenance cost share is 50.3%. Said percentages shall be adjusted to reflect actual flows, with such adjustments mutually agreed upon in writing following one year of operation of the Pure Water Monterey Project, and for each year thereafter.
 - 2. PCA's 4,320 acre-ft/year share, prorated monthly (360 acre-ft/month), shall have first priority should any curtailments of the 8,701 acre-ft/year of New Source Water takes place. Said priority shall commence upon completion and operation of any one of the New Source Water Facilities.
 - 3. Primary and secondary treatment of Agricultural Wash Water is managed by a separate agreement between Salinas and PCA, with Salinas making payments for aforesaid treatment. WRA will be responsible for only operations, maintenance, repair and replacement costs related to tertiary treatment of the volume of Agricultural Wash Water that is delivered to CSIP. Prior to completion and operation of the Pure Water Monterey Project, PCA agrees that 100% of the Agricultural Wash Water delivered to it by Salinas will be dedicated to CSIP.
 - 4. All New Source Waters shall be provided to CSIP, if desired by WRA, in a volume up to 50.3% of the total annual flow entering the Regional Treatment Plant from those sources, subject to the PCA first priority described in Section 4.02 (2) above. Flows not desired by

WRA may be utilized by PCA for the Pure Water Monterey Project, other purposes, or be discharged.

- (a) With the exception of any other New Source Waters the primary and secondary treatment costs of which are paid by others, WRA will be proportionately assessed for the incremental operation and maintenance costs of the influent pump station, primary treatment and secondary treatment of its portion of New Source Water flows actually delivered to tertiary treatment.
- (b) WRA will be responsible for incremental tertiary treatment operations, maintenance, repair and replacement costs related to the volume of New Source Waters that are delivered to CSIP.

4.03. Operations and Maintenance Treatment Cost Allocation.

Table	3.
-------	----

O&M Treatment Cost Allocation				
Source	Primary and Secondary	Tertiary	Pure Water Monterey Facilities	
Domestic W/W	Member Entities	WRA	PCA-Winter	
Ag Wash Water \$198 ²	Salinas (I.R)	WRA	PCA	
Blanco Drain \$74 ²	WRA/PCA (IR)	WRAt	PCA	
Rec/Tembladero Ditch \$77/\$91 ²	WRA/PCA (IR)	WRA	PCA	
Storm Water – Monterey \$69 ²	Monterey (IR)	WRA	PCA	
Salinas Pond Water Return Facilities \$83 ²	Intended User (IR)	WRA ^t	PCA	
Storm Water – Salinas \$69 ²	Salinas (IR)	WRA ^t	PCA	
Future WW in 2001 PCA Bound.	Member Entities	WRA ^t	PCA – Winter	
Future WW outside 2001 PCA Bound.	Member Entities	WRA (50% of	PCA – Winter (50% of	
		Flow if through	Flow if through SVRP; all	
		SVRP) [†]	flows otherwise	
MCWD – Delivery	Member Entities	MCWD	MCWD	
Interruptible Rate (IR)				

^tPCA – if through tertiary treatment; operations and maintenance and increased interest rate

²Interruptible rates are subject to change by PCA Board as described in Exhibit F, attached hereto and made a part hereof.

All treatment costs associated with the New Source Waters will be incorporated into the annual WRA budget process, as identified in Section VII. Annual Source Water costs will be based on the demand schedule provided by WRA to PCA; and PCA's Interruptible Rate Schedule for New Source Waters.

- 1. If any party utilizes tertiary treated water, other than water provided to PCA pursuant to Section 3.06, Incidental Uses, it shall be proportionately assessed by PCA for operation, maintenance, repair and replacement costs and increased interest costs, for the amount of water claimed and utilized from tertiary treatment facilities.
- 2. PCA agrees that it will cause MCWD, or itself, in use of its recycled water entitlements, to comply with all applicable requirements set forth in Contract No. 5-07-20-W1284, between the Bureau of Reclamation and WRA including, but not limited to, those contained in the Bureau Loan Agreement, all at PCA's or MCWD's sole cost and expense.

4.04. New Incremental and Interruptible Allocations - Phase II.

PCA and WRA plan to acquire and treat additional flows up to the Regional Treatment Facility's

capacity of 33,154 acre-ft/year, (hereinafter described as "Additional Flows"). Terms and conditions for accommodating Phase II flows will be subject to a separate future agreement.

4.05. Provisions for Drought Reserve.

PCA commits to produce no greater than 200 acre-ft/year, up to a total quantity of 1000 acre-ft of Pure Water Monterey product water for delivery to WMD for storage as a Drought Reserve.

- 1. PCA may, from time to time, request that WMD establish a Drought Reserve in excess of 1000 acre-ft.
- 2. During any fiscal year, PCA will first provide 3,500 acre-ft of Pure Water Monterey product water for delivery to WMD for urban water supply requirements before water is produced and delivered to the Drought Reserve.
- 3. In addition to water supplies provided for this Section IV, WRA may request additional irrigation water from PCA sources. When such a request is made, WMD may make available water from the Drought Reserve to meet urban water supply needs, allowing PCA to deliver additional source waters of an equal amount to WRA. In no instance shall PCA be required to reduce deliveries to WMD to make available additional irrigation water to WRA from its sources in an amount exceeding the balance available in the Drought Reserve.
- 4.06. Water Quality Enhancement Options.
 - 1. In order to protect against possible future recycled water quality deterioration, WRA desires that PCA evaluate and consider an expansion of the Pure Water Monterey 3,700 acre-ft/year advanced water treatment facility and construction of a pipeline. This potential expansion with a pipeline is to provide advanced treated recycled water to blend with the existing tertiary recycled water. This blending of water qualities will provide better water for growers in the CSIP project area.
 - (a) PCA shall complete an evaluation under California Environmental Quality Act (CEQA) within 180 days of the Effective Date of this Water Recycling Agreement, to determine whether the Expansion Project would result in a new significant adverse environmental effect or a substantial increase in an adverse environmental effect identified in the final EIR for the Pure Water Monterey Groundwater Replenishment Project.
 - (b) If after completion of the foresaid CEQA evaluation, PCA decides to approve the Expansion Project, PCA diligently pursue any available financial grant opportunities to design and construct the aforesaid advanced water treatment facility expansion and pipeline.
 - (c) WRA shall provide written notice to PCA when it desires the aforesaid advanced water treatment facility and pipeline to be operational and the amount of additional treatment capacity desired.
 - (d) If after completion of the foresaid CEQA evaluation, PCA decides to approve the Expansion Project, WRA and PCA shall negotiate a funding arrangement for the nongrant funded portion of the advanced water treatment facility expansion and pipeline. Related operation and maintenance costs to WRA shall be only incremental costs.
 - 2. If upon completion of CPUC CEQA review of the California-American Water Company proposed desalination facility, if the foresaid desalination facility is determined to be the selected alternative, PCA will allow California-American Water Company to finance, design, construct, own, and operate pipeline facilities connecting its proposed desalination facility to the tertiary treatment facility ponds in order to supply desalination facility basin return water to CSIP.

V. DELIVERY OF WATER TO WRA.

5.01. Quantity of water to be delivered to WRA SVRP.

Each year during the term of this contract, PCA shall deliver tertiary treated water from the SVRP to the Castroville Irrigation System. Subject to PCA's and MCWD's rights to the water as set forth in Article IV, the quantity so delivered shall include water provided in response to WRA's demand, determined as set forth in this Water Recycling Agreement.

5.02. WRA's demand schedule.

For water which will be delivered during the term of this Water Recycling Agreement, WRA shall, not later than November 1 each year, submit to PCA a schedule (see Exhibit C, attached hereto and made a part hereof) of requested water deliveries for the next calendar year. The schedule may request water deliveries of amounts up to amounts provided for WRA pursuant to Section 4.01 and 4.02 as WRA's demand. This demand schedule shall show the amount requested for each calendar month in the year.

5.03. WRA's demand.

It is understood and agreed that the wastewater committed by PCA, to which WRA is entitled pursuant to this Water Recycling Agreement, is dedicated for use by WRA for purposes of the CSIP as that seawater intrusion project as it may exist from time to time, and thereafter for direct application for such other non-urban agricultural irrigation projects in the Salinas Valley as WRA may develop for the purpose of curing seawater intrusion and other severe impacts of the groundwater imbalance in the Salinas Valley which threaten the viability of the groundwater basin as a water supply for agricultural and municipal uses, and for no other purpose. Where otherwise consistent with WRA's entitlement pursuant to this Water Recycling Agreement, WRA may use tertiary treated water produced by the SVRP for non-agricultural irrigation applications in the Salinas Valley only if PCA declines in writing to undertake such applications in the Salinas Valley; provided, however, PCA consents in writing to WRA's doing so, which consent shall not be unreasonably withheld. In no event shall PCA's obligation to provide secondary treated wastewater exceed the amount needed to produce the tertiary treated water to which WRA is entitled pursuant to this Water Recycling Agreement.

5.04. PCA'S duty to comply with WRA demand.

PCA shall supply the amount of water demanded in accordance with this Water Recycling Agreement and the demand schedule submitted by WRA, except when the SVRP and SRDF cannot produce enough water to satisfy the WRA's demand, PCA's demand, and MCWD's demand, all as limited by their respective entitlements, either because there is not enough watewater flowing into the PCA Regional Treatment Plant, or because, for reasons beyond the reasonable control of PCA, the SVRP is unable to operate for a sufficient period of time or the SVRP fails to produce enough water of the quality described in Section 6.01, "Water Quality," for use in the CSIP.

VI. MAINTENANCE OF WATER QUALITY.

6.01. Water Quality.

1. <u>General Water Quality:</u> PCA will produce tertiary treated water of a quality suitable for the irrigation of edible crops, such as artichokes, Brussels sprouts, and row crops (e.g., lettuce, cauliflower, broccoli, and celery) that do not require cooking prior to human consumption. All water produced and delivered to WRA shall meet all applicable standards of quality

prescribed by the State of California (including, but not limited to, the regulations promulgated by the State Health Department and set forth in the California Code of Regulations, Title 22), or by separate agreement of the parties, so that the water may be used for the purposes specified herein. The parties clarify in this section 6.01 their intent with regard to the required water quality and further agree that the SVRP has been designed to produce tertiary treated water for the irrigation of food crops where the recycled water comes in contact with the edible portions of the crop. The water quality which the SVRP was designed to produce is defined by the Recycled Water Criteria contained in Chapter 3 of Title 22 of the State of California's Administrative Code. In order to use this water for its intended purpose of irrigating food crops, the specifications contained in Water Reclamation Requirements Order No. 94-82, issued by the State of California Regional Water Quality Control Board, Central Coastal Region, on September 9, 1994, or as said water reclamation requirements may from time to time be revised and reissued by said Regional Water Quality Control Board, must be met. PCA will conduct a source control program to help provide reclaimed water of the best quality for food crop irrigation. However, the water quality specifications described in this Section 6.01 Water Quality constitute the sole and complete definition of the water quality that the SVRP will be expected to produce under this Water Recycling Agreement. PCA makes no representations as to the water quality the SVRP will produce with respect to other quality parameters not specifically set forth in the documents referred to in this Section 6.01.

6.02. Duty to monitor water quality: cessation in deliveries,

PCA will monitor the quality of water produced by the SVRP and SRDF, in accordance with the Reclaimed Water Monitoring Requirements contained in the Water Reclamation Requirements Order No. 94-82, referred to in Section 6.01, to ensure that the water delivered to the CSIP meets the quality standards set forth in Section 6.01. PCA may perform additional monitoring which it deems appropriate. The parties understand that it may not be possible for the water produced by the SVRP and SRDF to meet these standards 100% of the time, and that water quality will vary from time to time. During periods when the PCA determines that water quality does not meet the standards, set forth in Section 6.01, flows will not be sent to the CSIP. For those occasions, the CSIP's backup water source will be the SRDF water if available or the CSIP's supplemental wells. PCA will provide special attention to minimize any such events to the extent that it has control over them. PCA will notify WRA promptly, whenever it appears that there is or will soon be a change in the quality of water produced by the SVRP that may adversely affect the agricultural users of the water.

6.03. Regulations to protect water quality.

PCA will, to the extent feasible, enact reasonable and appropriate regulations governing the kinds of wastes and other materials that may be discharged into the sewerage system and/or agricultural wash water system, in order to protect the quality of water ultimately produced.

6.04. Water Quality and Operations Committee.

The Parties shall maintain a Water Quality and Operations Committee (Committee) whose membership shall consist of representatives of the parties and the areas receiving CSIP water. The Committee is not subject to the provisions of the Open Meeting Law (Cal. Gov't Code section 6250, et seq.) The Committee shall have access to and shall share all pertinent information concerning the operation of the SVRP, SRDF, and CSIP, the quality of water produced, and the needs of the growers receiving water, in order to discuss and make recommendations for maintaining or improving the SVRP, SRDF, and CSIP so that it will continue to meet the needs of the agricultural operations

served. No private sector member of the Committee and no grower or private sector user of water shall be required to disclose proprietary information in connection with the Committee's work, except as may be required by governing law or regulation. The SRDF Water Quality Monitoring Plan shall be presented to the Water Quality and Operations Committee at least once per year or prior to any significant changes, for committee review and input. Any financial changes approved by the Water Quality and Operations Committee at this meeting will be submitted to the WRA Board of Director's for approval and the approved Board Resolution will be forwarded to PCA for any budget modifications.

VII. PAYMENTS BY WRA TO PCA

7.01. Consideration paid by WRA.

As consideration for the water provided and the other obligations performed by PCA under this Water Recycling Agreement, WRA shall make payments to PCA pursuant to this Water Recycling Agreement.

7.02. Amounts to be paid.

The consideration paid by WRA shall be the dollar amount that equals 100% of the annual costs reasonably and necessarily incurred in connection with the SVRP, CSIP, and SRDF (including both direct and indirect expenses), and no more. PCA will make all reasonable efforts to stay within, or below, the amounts budgeted for the SVRP, CSIP and SRDF. All budgetary savings will be applied to the following year's budgets. The annual costs include the following:

- 1. Amortization of any loans (USBR, SWRCB, etc.), bonds, certificates of participation, and any interim or other financing;
- 2. The SVRP, CSIP, and SRDF operation, maintenance, repair, replacement, power, capitalized equipment, capital improvements, contingency, and reserve costs as are contained in the budgets for the SVRP, CSIP and the SRDF. PCA will make all reasonable efforts to stay within the total budget amounts for these budgets, but will have the latitude to incur variances between individual budget accounts, so long as the total budget amounts are not exceeded.
- 3. Unused funds received from WRA will be placed by PCA in an interest bearing account. All interest earned on WRA accounts will be applied for WRA's benefit on a monthly basis. PCA will adjust the second semi-annual billing and year-end reconciliation statement by any interest earned on WRA accounts. PCA shall be reimbursed by WRA for the reasonable loss of interest earnings on PCA's monies used to cover any SVRP or CSIP operations and maintenance costs until such costs are reimbursed by WRA.

7.03. Payment schedule.

PCA will establish separate accounts for the SVRP, SRDF, and the CSIP. Each year, WRA will fund each account by June 30 with a payment that adjusts seed monies to 50% of the projected operations, maintenance, capital outlay, and reserve expenses for the next fiscal year. Thereafter, PCA will record into this account the payments noted in Section 7.04 based on cost share estimates determined pursuant to Section 4.02 (1) herein.

- 1. WRA will make six payments to PCA each year as follows:
 - (a) Thirty (30) days before the date that PCA's annual payment on the USBR loan for the SVRP is due, WRA will pay an amount equal to the then current payment on the USBR loan;

- (b) Thirty (30) days before the date that PCA's annual payment on the SWRCB loan for the SVRP is due, WRA will pay an amount equal to the current payment on the SWRCB loan;
- (c) Thirty (30) days before the date the PCA's annual payment on the SWRCB loan for the New Source Water Facilities/Pure Water Monterey Project is due, WRA will pay an amount equal to WRA's proportional share of capital costs for the New Source Water Facilities, as provided in Exhibit H attached hereto and made a part hereof.
- (d) On March 1, WRA will pay the SVRP, CSIP, and SRDF operation, maintenance, reserve, and all other SVRP and CSIP costs for the immediately preceding July-December that are allocable to WRA and payable hereunder;
- (e) On September 1, WRA will pay the SVRP, CSIP, and SRDF operation, maintenance, reserve, and all other costs budgeted for the immediately preceding January-June that are in excess of the payments received allocable to WRA and payable hereunder. All surplus costs related to SVRP, CSIP, and SRDF operations will be processed in accordance with Section 7.02.
- (f) On March 1, WRA pay PCA the proportional share of the amortized replacement/renewal costs of the New Source Water Facilities as identified in Exhibit I, attached hereto and made a part hereof.
- (g) Debt Service and Amortization schedules are shown in Exhibit H.

7.04. Payment procedures.

- 1. At least thirty (30) days before the specified due dates for the payments under Section 7.03, PCA will send to WRA a request for payment, indicating the amount due, the date payment from WRA is due, and the nature of the payment.
- 2. Payment requests for operation, maintenance, reserve and other costs shall contain the following information: Expenditure reports for SVRP, CSIP, and SRDF that describe year-to-date (YTD) expenditure, YTD encumbrance, and YTD total, for each account along with the YTD total for all accounts, including indirect costs and contingency, and the percentage used YTD of the budgeted amount. Back-up information (original paid invoices, payroll records, time sheets, and vehicle costs) will be maintained by PCA to support each direct charge.
- 3. WRA shall promptly review all requests for payments and shall approve for payment such requests or portion thereof that are in conformity with this Water Recycling Agreement and shall promptly submit such approved requests to the County Auditor for payment. WRA shall promptly notify PCA of any disputed payment request. Disputes over payments shall be resolved as provided in Article XIII of this Water Recycling Agreement.
- 7.05. Reduction or termination of water delivery for lack of payment; judicial remedies.
 - 1. Notwithstanding anything to the contrary in this Water Recycling Agreement (except for the provisions of paragraph (b), below), if WRA should fail to make any payment to PCA required under this Water Recycling Agreement for a period of ninety (90) days or more after the due date, then PCA, upon fifteen (15) days' written notice to WRA, may act to reduce or terminate the delivery and/or distribution of tertiary treated water from the SVRP and/or discontinue operation of the CSIP. Such reduction or termination in the delivery of water or discontinuance of operations of CSIP may continue only for such period of time as payments from WRA to PCA remain delinquent, and may continue even though dispute resolution procedures pursuant to Article XIII are pending between the parties, unless WRA makes full payment of the amount in question, under protest, as provided in Section 13.01.

- 2. For as long as any notes or bonds issued to finance the SVRP or CSIP are outstanding, PCA will not exercise its right to terminate or reduce the water supply of the tertiary treated water to the CSIP, except that PCA may reduce the water supply to the extent that the failure of WRA leaves PCA without all necessary and sufficient funds to operate the SVRP.
- 3. In addition, if WRA should fail to make any payment to PCA required under this Water Recycling Agreement for a period of ninety (90) days or more after the due date, the PCA shall have the right to seek any appropriate judicial relief, at law or in equity, for such default. Such relief may include, but need not be limited to, damages, injunctive relief, and the appointment of a receiver.

7.06. Prior approval of budget by WRA.

- 1. Each year, in accordance with its normal budgeting schedule, PCA will adopt budgets for the operation of the SVRP, CSIP, and SRDF, in advance of the operating year for which the budgets are proposed, including therein all costs to be paid by WRA. In preparing these budgets, the demand schedule approved by WRA's General Manager will be provided to PCA by November 1 of each year. Within six (6) weeks of receipt of the demand schedule, PCA will provide WRA with a preliminary budget estimate. Any additional costs associated with the New Source Waters will be calculated based on the demand schedule and PCA's Interruptible Rate; and will be included as a separate line item in the SVRP, CSIP, or SRDF budget. Upon receipt of the preliminary budget, both agencies will agree to meet in a timely manner sufficient to enable review and approval by both agencies' respective governing boards.
- 2. Once the budgets are approved by WRA's Board of Directors, WRA's Board of Supervisors, and PCA's Board of Directors, PCA will not modify the budgets so as to exceed the amount of expenditures approved by WRA without first submitting the proposed changes to WRA's Board of Directors and WRA's Board of Supervisors for review and approval, except as provided for in Section 7.11. In addition, both agencies will provide advance notification of any proposed budgetary modifications that will be considered by each agency's respective board.
- 3. In the event WRA's revenues to fund the CSIP, SVRP, and SRDF budgets fall below projected amounts, WRA may request that PCA revise the operations to incur lower operating costs to stay within the revised revenue projections.

7.07. SWRCB as third party beneficiary.

The State Water Resources Control Board (SWRCB) shall be a third party beneficiary of the WRA's payment obligations under this Article VII, to the extent that the WRA's payment obligations are intended to provide funds to PCA to repay the loan to the PCA from the SWRCB. In the event of default by the PCA in the repayment of said loan, the SWRCB may enforce the loan repayment obligations against the WRA, and to the extent that the WRA makes payment to the SWRCB, the WRA shall be relieved of its corresponding obligation to the PCA. If, through no fault of the WRA, the PCA defaults in its repayment obligation to the SWRCB, and the WRA thereby becomes obligated to pay any penalties, extra charges, or amounts in excess of the basic principal and interest on the loan, then PCA shall reimburse WRA for any such amounts paid by WRA.

7.08. Priorities of payment.

Notwithstanding anything to the contrary contained herein, WRA's obligations to make payments with respect to the SVRP, CSIP, and SRDF shall be prioritized as follows, and the obligations in each category shall be subordinate to the obligations in each prior category, shall be on a parity

with all other obligations in its category, and shall be senior to the obligations in each subsequent category:

- 1. Operation and maintenance costs for the SVRP, CSIP, and SRDF;
- 2. Debt service on obligations (other than USBR or SWRCB loans for the SVRP, CSIP, and the New Source Water Facilities) incurred to finance the SVRP, CSIP, and New Source Water Facilities, payments to replenish any reserve fund for such obligations, and payments to any provider of credit enhancement for such obligations;
- 3. Debt service on the USBR and SWRCB loans for the SVRP;
- 4. Debt service on the New Source Water Facilities;
- 5. Repair, replacement, capital improvements, capitalized equipment, contingency and reserve costs of the SVRP, CSIP, and SRDF (to the extent the same do not constitute operation and maintenance costs);
- 6. WRA's share of the amortized replacement/renewal costs of the New Source Water Facilities as identified in Exhibit I.

7.09. Application of loan payments by PCA.

All payments made by WRA to PCA for the repayment of the USBR or SWRCB SVRP loans shall be used for such repayment. Upon termination of any loan agreement, any unused funds returned to PCA and retained by PCA shall be returned to WRA within 60 days from the date of the approved PCA audit for the fiscal year in which the agreement was terminated all in accordance with Section 9.01 and 9.02 herein.

7.10. Unanticipated events/Emergency situations

- 1. <u>Unanticipated events.</u> Circumstances or events may arise which were not anticipated in either the scopes of services or the budgets for SVRP, CSIP, and SRDF. In this case, PCA will submit its plans for addressing such circumstances or events, including justification and estimated amount of expenditures, to the Water Quality and Operations Committee, formed in accordance with Section 6.04, for its review and recommendations. Before proceeding with those plans, WRA must first give its written approval to incur any additional costs associated therewith.
- 2. <u>Emergency situations.</u> If the event or circumstance constitutes an emergency situation which threatens health and safety, damage to property, or injury to persons, PCA will act as promptly and as efficiently as possible to mitigate the situation without waiting for approval by the Water Quality and Operations Committee or WRA. PCA will advise the Water Quality and Operations Committee and WRA as soon thereafter as possible of the mitigating actions taken and of any further action that may be necessary.

7.11. Decision-making authority.

In order to provide for the smooth and efficient operation of SVRP, CSIP, and SRDF, PCA will have the full authority to make and implement decisions with regard to activities and expenditures for these purposes without prior approval of WRA. All such activities shall be within the scope of services described in Exhibit B "Operation and Maintenance of SVRP, CSIP, SRDF, and New Source Water Facilities", respectively. All such expenditures shall be within the overall budget amounts for the SVRP, CSIP and SRDF.

VIII. ACCOUNTING SYSTEM, REPORTS

8.01. Accounting system.

PCA will maintain an accounting system that is in conformity with generally accepted accounting principles (GAAP) and will allow for the segregation and tracking of all revenues and direct costs related to the SVRP, CSIP, and SRDF. The accounting system shall properly allocate costs to the SVRP, CSIP, and SRDF and to PCA's other activities that are not subject to reimbursement by WRA under this Water Recycling Agreement. The accounting system shall provide the ability to adequately identify indirect cost centers and establish allocation factors to assign indirect costs proportionally to the CSIP, SVRP, SRDF, and the Interruptible Rate for New Source Waters. Cost accounting, including any overhead distributions, shall be in accordance with GAAP. PCA shall make the details of such system available to or known to WRA and/or to WRA's auditor, at WRA's request.

8.02. Financial reports.

By the last day of each month PCA will provide to WRA a report on expenditures made during the previous month and year-to-date, for each line item in the SVRP, CSIP, and SRDF budgets. These reports will identify any line items which vary appreciably from anticipated budget to date. PCA will provide WRA an annual report of WRA's proportional share of Reserves retained in the Pure Water Monterey Fund for New Source Water Facilities. This report will be provided by PCA to WRA by September 30 of each year; and include WRA's deposits made to the Repair/Renewal Reserve, proportional interest earned, and the proportional share of any replacement/renewal costs.

8.03. Direct and indirect costs.

- 1. Direct costs of the SVRP, CSIP and SRDF are costs which can be tracked as costs of these particular activities through invoices, time cards, record keeping systems, and other records that specifically allocate a cost to these activities. Indirect costs are all other costs incurred by PCA in order to manage, maintain, support, and operate the SVRP or the CSIP.
- 2. PCA shall implement the accounting system described in Section 8.01 to uniformly identify and allocate all direct and indirect costs for the SVRP and the CSIP and for all the PCA's other activities. PCA shall identify the specific functions that are typically considered administrative or support in nature. These functions or departments shall include Human Resources, Finance, Administration, Information Technology, and Safety. The annual budgeted costs of these functions will be allocated proportionally to all operational activities based on a percentage relational to the services provided to SVRP, CSIP, SRDF and all other PCA activities. A sample overview is provided in Exhibit J, attached hereto and made a part hereof. The budgets associated with the administrative or support functions will not be directly charged to the SVRP, SRDF, or CSIP. PCA will make reasonable efforts to maximize the extent to which costs to be paid by WRA can be identified as direct costs rather than as indirect costs.
- 3. For purposes of allocating indirect costs, PCA will not include as a direct cost the debt service (principal and interest) on the loans obtained for the project.
- 4. PCA and WRA retain the right to transition from the cost allocation plan identified in 8.02 (b) to a cost allocation model that is compliant with the Office of Management and Budget (OMB) Circular A-87 Cost Principles for State, Local, and Indian Tribe Governments or a subsequent revision. Any cost allocation subject to this provision shall be accompanied by a Certificate of Cost Allocation Plan and be in compliance with Title 2 CFR, Part 200. All indirect costs charged to functional activities will be applied consistently with the results of

this plan to ensure equity between costs centers and conformance with OMB standards; provided, however, no allocation of any indirect cost unrelated to SVRP, CSIP or SRDF shall be made.

- 5. If PCA elects to engage a consultant to comply with OMB Circular A-87, PCA shall be primarily responsible for the costs associated with the Plan, and WRA shall pay for a portion of the plan consistent with the current year indirect cost allocation factor. If WRA requests that PCA engage a consultant to comply with OMB Circular A-87, the costs associated with the plan shall be divided equally.
- 6. Implementation of an OMB Circular A-87 Cost Allocation Plan will become effective in the new fiscal year immediately following the cost allocation study.
- 7. All OMB Circular A-87 Cost Plans will be subject to updates as consistent with industry standards. All costs associated with a consultant retained to update the cost allocation plan shall be paid in accordance with Section 7.02 (e).

8.04. Annual audit.

The accounting for WRA's funds will be subject to PCA's Annual Audit. Information relating to WRA's activities will be classified as Restricted in PCA's Comprehensive Annual Financial Report (CAFR). This Restricted classification will remain in effect through the term of this agreement, unless there are any new Governmental Accounting Standards Board (GASB) pronouncements or auditor comments that require a change in classification. A copy of PCA's CAFR will be provided to WRA by January following the close of the subsequent fiscal year.

8.05. Right to inspect and audit records.

WRA shall have the right to inspect PCA's records pertaining to the SVRP, CSIP, and SRDF, upon reasonable advance notice. WRA shall also have the right to audit PCA's records pertaining to the SVRP, CSIP, and SRDF, or to have them audited by an auditor selected by WRA, at WRA's sole cost and expense. Such audit may be performed at any time during regular business hours, upon the giving of reasonable advance notice. If the audit shows that PCA has overcharged annual costs to WRA by more than five percent, PCA shall promptly on demand by WRA reimburse WRA for the cost of the audit.

8.06. Reimbursement for overcharge or undercharge.

If any audit shows that an undercharge or an overcharge has occurred, each agency will have 90 days to comply with the audit findings.

IX. REPAIR AND MODIFICATION OF SVRP, CSIP, SRDF, AND NEW SOURCE WATER FACILITIES AND RELATED FACILITIES.

9.01. SVRP, CSIP, and SRDF Reserve Funds.

PCA shall at all times during the term of this Water Recycling Agreement accumulate and maintain a reclamation reserve fund for SVRP, CSIP, and SRDF. The loan reserve fund will be at a minimum in accordance with each loan agreement listed in Exhibit G. The exact amount for each operating reserve shall be determined in the annual budget process. If insurance or self-insurance monies are available for any projects that are intended to be funded by reserves, the insurance monies will be used prior to accessing reserve funds. Upon termination of any agreement, any unused funds retained by PCA shall be returned to WRA within 60 days from the date of the approved PCA audit for the Fiscal Year in which the agreement was terminated.

9.02. Replacement and Renewal Costs Associated with New Source Water Facilities.

WRA shall pay PCA the proportional share of amortized capital renewal costs associated with the New Source Water Facilities. WRA's annual contribution is provided in Exhibit I. All funds received from WRA will be placed in the Pure Water Monterey Fund's Renewal and Replacement Reserve in accordance with Section VI. All funds will be held in the Reserve for the purpose of funding capital outlay projects for the New Source Water Facilities; assisting in meeting any fiscal sustainability plan requirements for the Clean Water State Revolving Loans; and maintaining a proportional share of the Loan's debt covenant ratio. At the completion of the thirty-year loan cycle, WRA and PCA will develop a long-term Capital Improvement Plan, which includes establishing an appropriate level of Renewal and Replacement reserves. Any WRA funds that are held in Reserves in excess of the Capital Improvement Plan will be refunded within ninety (90) days of the Plan's establishment.

9.03. Expansion of treatment plant capacity.

PCA may expand the Regional Treatment Facility above the current capacity of 29.6 MGD and may construct additional reclamation facilities, at its sole cost and expense and without receiving the consent of WRA. Any increases in capacity and any additional reclamation facilities so constructed shall be used at PCA's discretion, provided, however, that PCA is precluded from providing water to customers within WRA's water supply jurisdiction.

9.04. Interruptions of service.

- 1. No work of construction, remodeling, renovation, replacement, repairs, addition, or expansion authorized under this Water Recycling Agreement and performed on the SVRP, CSIP and SRDF shall, either before, during, or after such work, interfere with, interrupt, or reduce the delivery of tertiary treated water and river water to WRA under this Water Recycling Agreement, except that minor interferences, interruptions, or reductions shall be allowed when necessary, unavoidable, or beyond the control of PCA.
- 2. PCA shall schedule its planned maintenance activities on the SVRP, CSIP and SRDF to minimize interruption of distribution of tertiary, river and/or well water by the CSIP. Unscheduled work to perform repairs or maintenance will be performed in the manner deemed by PCA to have the least impact on the distribution of tertiary, river and/or well water by the CSIP. In case of any interruption of service, PCA shall give notice in the same manner as required by Section 3.07.

9.05. Duty to apply insurance proceeds.

If either party recovers any insurance proceeds on account of loss or damage to the SVRP, CSIP, or SRDF, such proceeds shall be applied to repair or replace the damaged portion of the SVRP, CSIP, or SRDF, and not otherwise. If either party is self-insured and any loss or damage occurs that would have been covered by insurance otherwise required to be maintained by such party under this Water Recycling Agreement, then such party shall provide the funds that would have been 'recovered had the party been insured and shall apply the funds to repair or replace the damaged portion of the SVRP, CSIP, or SRDF, or SRDF.

9.06. Payment of uninsured losses.

If the SVRP, CSIP, or SRDF is damaged or destroyed during the term of this Water Recycling Agreement, by other than an intentional or willful misconduct of a PCA employee, and the amount of available insurance and self-insurance monies plus replacement reserves is insufficient to repair or replace the damage, then WRA shall pay the balance necessary to restore these facilities to their condition prior to the damage.

9.07. New Water Recycling Agreement on replacement of SVRP.

Should all, or part of, the SVRP, CSIP, or SRDF require replacement, whether from insurance or self-insurance monies or from the reserve for replacement, the Parties will in a good faith attempt to negotiate and reach agreement on a new Water Recycling Agreement on terms substantially in accord with the present Water Recycling Agreement.

X. INDEMNIFICATION.

10.01. Indemnification of WRA.

PCA shall immediately indemnify, defend, and hold harmless WRA and the County of Monterey (the "County"), their officers, agents, and employees, from and against any and all suits, claims, liabilities, and losses whatsoever against WRA or the County (including damages to property and injuries to or death of persons, court costs, and reasonable attorneys' fees) occurring or resulting to any and all persons, firms or corporations furnishing or supplying work, services, materials, or supplies in connection with the performance of this Water Recycling Agreement, and from any and all suits, claims, liabilities, and losses occurring or resulting to any person, firm, or corporation for damage, injury, or death arising out of or connected with the PCA's performance of its obligations pursuant to this Water Recycling Agreement caused in whole or in part by any negligent act or omission of PCA, any subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, except where caused by the active negligence, sole negligence or willful misconduct of WRA.

10.02. Indemnification of PCA.

WRA shall immediately indemnify, defend, and hold harmless PCA, its officers, agents, and employees, from and against any and all suits, claims, liabilities, and losses whatsoever against PCA (including damages to property and injuries to or death of persons, court costs, and reasonable attorneys' fees) occurring or resulting to any and all persons, firms or corporations furnishing or supplying work, services, materials, or supplies in connection with the performance of this Water Recycling Agreement, and from any and all suits claims, liabilities, and losses occurring or resulting to any person, firm, or corporation for damage, injury, or death arising out of or connected with the WRA's performance of its obligations pursuant to this Water Recycling Agreement caused in whole or in part by any negligent act or omission of WRA, any subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, except where caused by the active negligence, sole negligence or willful misconduct of PCA.

10.03. Procedure for Indemnification.

- 1. If any legal or administrative proceedings are instituted, or any claim or demand is asserted, by any third party which may give rise to any damage, liability loss or cost or expense with respect to which either party has agreed to indemnify the other party in this agreement, then the indemnified party shall give the indemnifying party written notice of the institution of such proceedings, or the assertion of such claim or demand, promptly after the indemnified party first becomes aware thereof. However, any failure by the indemnified party to give such notice on such prompt basis shall not affect any of its rights to indemnifying party to defend such proceeding.
- 2. The indemnifying party shall have the right, at its option and at its own expense, to utilize counsel of its choice in connection with such proceeding, claim or demand, subject to the

approval of the indemnified party. The indemnifying party shall also have the right to defend against, negotiate with respect to, settle or otherwise deal with such proceeding, claim or demand. However, no settlement of such proceeding, claim or demand shall be made without the prior written consent of the indemnified party, which consent shall not be unreasonably withheld or delayed. The indemnified party may participate in any such proceeding with counsel of its choice at its own expense.

- 3. In the event, or to the extent, the indemnifying party elects not to, or fails to, defend such proceeding, claim or demand and the indemnified party defends against, settles or otherwise deals with any such proceeding, claim or demand, any settlement thereof may be made without the consent of the indemnifying party if it is given written notice of the material terms and conditions of such settlement at least ten days before a binding agreement with respect to such settlement is executed.
- 4. Each of the Parties agrees to cooperate fully with each other in connection with the defense, negotiation or settlement or any such proceeding, claim or demand.

10.04. Payment of indemnified claims.

The indemnifying party shall forthwith pay all of the sums owing to or on behalf of the indemnified party, upon the happening of any of the following events:

- 1. Upon the rendition of a final judgment or award with respect to any proceeding described in Section 10.03, above, by a court, arbitration board or administrative agency of competent jurisdiction and upon the expiration of the time in which an appeal therefrom may be made; or
- 2. Upon the making of a settlement of such proceeding, claim or demand; or
- 3. Upon the parties' making of a mutually binding agreement with respect to each separate matter indemnified thereunder.

10.05. Contribution in the event of shared liability.

In the event any proceeding, claim or demand described in Section 10.03 is brought, in which allegations of fault are made against both the parties, the extent of indemnification shall be determined in accordance with the agreement of the parties, or, if there is no agreement, then in accordance with the findings of the court as to the relative contribution by each of the parties to the damage suffered by the party seeking indemnity with respect to such proceedings.

10.06. Exclusion from O&M costs.

Amounts payable by either party as indemnification shall not be included in the operations and maintenance costs of the SVRP, CSIP, SRDF, and New Source Water Facilities.

- 1. The indemnifying party shall have the right, at its option and at its own expense, to utilize counsel of its choice in connection with such proceeding, claim or demand, subject to the approval of the indemnified party, which approval shall not be unreasonably withheld or delayed. The indemnifying party shall also have the right to defend against, negotiate with respect to, settle or otherwise deal with such proceeding, claim or demand. However, no settlement of such proceeding, claim or demand shall be made without the prior written consent of the indemnified party, which consent shall not be unreasonably withheld or delayed, unless, pursuant to the terms and conditions of such settlement, the indemnified party is released from any liability or other exposure with respect to such proceeding, claim or demand. The indemnified party may participate in any such proceeding with counsel of its choice at its own expense.
- 2. In the event, or to the extent, the indemnifying party elects not to, or fails to, defend such proceeding, claim or demand and the indemnified party defends against, settles or otherwise

deals with any such proceeding, claim or demand, any settlement thereof may be made without the consent of the indemnifying party if it is given written notice of the material terms and conditions of such settlement at least ten days before a binding agreement with respect to such settlement is reached.

3. Each of the Parties agrees to cooperate fully with each other in connection with the defense, negotiation or settlement of any such proceeding, claim or demand.

XI. INSURANCE.

11.01. General insurance requirements.

Without limiting PCA's duty to indemnify, PCA shall maintain in effect throughout the term of this Water Recycling Agreement a policy or policies of insurance meeting the requirements hereinafter set forth. All such insurance required by this article shall meet the following requirements:

- 1. Each policy shall be with a company authorized by law to transact insurance business in the State of California, and shall be written on an occurrence form.
- 2. Each policy shall provide that both Parties shall be given notice in writing at least thirty days in advance of any change, cancellation or non-renewal thereof.
- 3. Except with respect to workers compensation insurance, each policy shall provide an endorsement naming both Parties and its officers, agents and employees as additional insureds, or additional insureds, as applicable, and shall further provide that such insurance is primary to any other insurance maintained by the WRA.
- 4. PCA shall require each general contractor and sub-contractor performing construction work under this agreement, or shall be accompanied by a certificate of insurance showing that the contractor or sub-contractor has identical insurance coverage or an appropriate substitute for such coverage approved by the parties hereto.

11.02. Commercial general liability insurance.

WRA and PCA shall maintain (and be named insured under) commercial general liability insurance covering all operations under this Water Recycling Agreement, with such coverages as the parties may agree upon from time to time. The County of Monterey shall be named as an additional insured. PCA will pay up to 40% of the annual cost, for the term of this Water Recycling Agreement. In addition, should this Water Recycling Agreement be terminated by the parties under Section XII with regard to the operation of the CSIP, the obligation of the PCA to pay for such insurance regarding the CSIP shall be accordingly reduced.

11.03. Motor vehicle insurance.

PCA shall maintain insurance covering all motor vehicles (including owned and non-owned) used in providing services under this Water Recycling Agreement, with a combined single limit of not less than \$2,000,000.

11.04. Property insurance.

1. PCA shall maintain insurance covering the SVRP against loss or damage due to fire and other perils to the extent that such insurance is readily and practically available and within available funds for the SVRP. The amount of the insurance shall not be less than the then-current replacement cost of the SVRP, without depreciation. Insurance coverage for the SVRP under this section shall be reviewed and approved by WRA, which shall not unreasonably withhold or delay its approval. PCA shall provide WRA with a copy of the

insurance policy and shall give WRA thirty (30) days' advance notice of any cancellation or proposed change in the insurance required by this section, and any such change shall be subject to review and approval by WRA.

2. WRA shall maintain insurance covering the CSIP and SRDF against loss or damage due to fire and other perils to the extent that such insurance is readily and practically available and within available funds for the SRDF. The amount of the insurance shall be not less than the then-current replacement cost of the CSIP and SRDF, without depreciation. Insurance coverage for the CSIP and SRDF shall be reviewed and approved by PCA, which shall not unreasonably withhold or delay its approval. WRA shall provide PCA with a copy of the insurance policy and shall give PCA thirty (30) days' advance notice of any cancellation or proposed change in the insurance required by this section, and any such change shall be subject to review and approval by PCA.

11.05. Workers' compensation insurance.

PCA shall maintain a workers' compensation plan covering all of its employees as required by Labor Code Sec 3700, either (a) through workers' compensation insurance issued by an insurance company, with coverage meeting the statutory limits and with a minimum of \$100,000 per accident for employer's liability, or (b) through a plan of self-insurance certified by the State Director of Industrial Relations, with equivalent coverage. If PCA elects to be self-insured, the certificate of insurance otherwise required by this Water Recycling Agreement shall be replaced with a consent to self-insure issued by the State Director of Industrial Relations.

11.06. Certificate of insurance.

PCA shall file certificates of insurance with the WRA and with the Monterey County Risk Management Division, showing that PCA has in effect the insurance required by this agreement. PCA shall file a new or amended certificate promptly after any change is made in any insurance policy which would alter the information on the certificate then on file.

11.07. Self-insurance.

Each Party may, instead of obtaining the insurance required of it by this Water Recycling Agreement, elect to be self-insured or to participate in a self-insurance pool, provided the other Party first gives its written consent, which will not be unreasonably withheld or delayed. The Parties shall enter into a separate written memorandum of understanding specifying the proportionate amount or share of such self-insurance allocated to SVRP, CSIP, and SRDF.

11.08. Insurance costs.

Insurance expenses and equivalent self-insurance expenses are CSIP/SVRP costs, except for: costs excludable under indemnification, PCA's share of the liability insurance premium costs under Section 11.02; and costs of PCA's workers' compensation insurance covering work outside the scope of this Water Recycling Agreement. All deductibles under Section 11.02 liability insurance and Section 11.04 property insurance, as well as any self-insured retention under excess insurance held by WRA, PCA, or Monterey County will be borne by WRA. All damages, liabilities, losses and costs, including fines and penalties, that are not covered under a policy of insurance, a self-insurance pool, or other self-insurance program shall be borne by the Parties according to this Water Recycling Agreement.

11.09. Periodic increases in coverage requirements.

The coverage limits stated herein in dollar values shall be adjusted upwards by 15% for every 15% increase in the consumer price index for all urban consumers in the San Francisco Bay area over the

level in effect on the date of this agreement or the previous level, or, if such index is not available, for any substitute or replacement index. Either party shall notify the other when such increase is requested or made.

XII. TERM OF AGREEMENT.

12.01. Term of Agreement.

This Water Recycling Agreement shall become effective on the date hereinabove first made and entered into and shall terminate on the latter of (i) December 31 of year 2045 or (ii) December 31 of the year in which full repayment of debt instruments utilized for financing of New Source Water Facilities is made; subject to any extensions provided for in Section 12.02.

12.02. Automatic extension.

This Water Recycling Agreement shall be automatically extended and renewed for additional 10-year periods unless one Party provides the other Party written notice by December 31st of the year preceding the year of expiration that an extension is not to occur.

12.03. Conditions of agreement during term.

All the terms of this Water Recycling Agreement shall remain in effect during any extended term, except as otherwise provided in this agreement.

12.04. Rights on Termination.

- 1. Upon any termination of this Water Recycling Agreement, WRA shall have the continuing right to receive from PCA the same quantity of incoming wastewater flow as originating in the Salinas Valley. WRA shall provide all facilities for treating the water beyond secondary treatment level at its sole cost and expense or through a subsequent cooperative agreement with PCA or any other entity. Upon any termination of this Water Recycling Agreement, PCA shall have the continuing right to receive the same quantity of tertiary treated water as PCA was or would have been entitled to receive during any term of this Water Recycling Agreement.
- 2. Upon written notice to PCA within 90 days of termination of this Water Recycling Agreement, WRA shall, at its sole discretion and to the extent permitted by law, have the option to lease or purchase (as determined by PCA) the SVRP and the land upon which it sits, and thereafter to operate, maintain and otherwise be responsible for the SVRP at WRA's sole cost and liability. Upon notice from WRA of its intent to exercise this option, PCA shall, within 90 days, inform WRA whether the land and the SVRP must be acquired, leased, or a combination thereof. WRA thereafter shall indemnify and defend PCA with respect to such facilities and lands, as set forth in a sale or lease agreement.
- 3. If the parties are unable to agree on the fair market value for the sale or lease of the SVRP and of the underlying land, that dispute shall be decided in accordance with the alternative dispute resolution provisions set forth in Article XIII.
- 4. PCA's rights to tertiary treated water in accordance with this Water Recycling Agreement shall also survive cancellation, regardless of whether WRA exercises the option pursuant to Section 12.04(2).

XIII. DISPUTE RESOLUTION.

13.01. Dispute resolution procedure.

If any dispute arises between the Parties as to the proper interpretation or application of this Agreement, and/or the proper operation of the facilities, the Parties shall first seek to resolve the dispute in accordance with this Article, and the Parties must proceed through arbitration under this Article before filing any court action as set out below.

13.02. Duty to meet and confer.

If any dispute under this Agreement arises, the Parties shall first meet and confer, in an attempt to resolve the matter between themselves. Each Party shall make all reasonable efforts to provide to the other Party all the information that the Party has in its possession that is relevant to the dispute, so that both Parties will have ample information with which to reach a decision.

13.03. Mediation and Arbitration.

- 1. If the dispute is not resolved by meeting and conferring within 30 days of commencing that process, the matter may be submitted to mediation. The terms of and process for the mediation shall be set by the Parties and the mediator. The Parties agree that they may, by or through consensus, elect to convert the mediation into arbitration.
- 2. If the dispute is not resolved by meeting and conferring, and mediation is not chosen or not implemented by the parties, or is unsuccessful, the Parties shall submit the matter to arbitration. In that event the Parties will jointly select a single arbitrator, or, if the Parties are unable to agree, they shall each select an arbitrator, and the matter shall be handled by two arbitrators. The two arbitrators may, if they deem it appropriate and warranted by the nature and significance of the dispute, themselves select a third arbitrator. Any person selected as an arbitrator shall be a qualified professional with expertise in the area that is the subject of the dispute, unless the Parties otherwise agree. The arbitration shall be conducted in accordance with the rules of the American Arbitration Association. The decision of the arbitrator or arbitrators shall be binding, unless within 30 days after issuance of the arbitrator's written decision, either party files an action in court.

XIV. ADDITIONAL RESPONSIBILITIES.

14.01. Compliance with water reclamation requirements for the CSIP.

PCA shall be responsible for compliance with all of the requirements contained in the "Recycled Water User Requirements for Monterey County Water Resources Agency (User), Castroville Seawater Intrusion Project, Monterey County," Order No. 97-52, issued by the State of California Regional Water Quality Control Board, Central Coast Region, on September 5, 1997, or as said Order may from time to time be revised and re-issued by the Regional Water Quality Control Board. PCA's responsibilities shall be limited to complying with the Water Reclamation Requirements for the SVRP, as described in Section 6.01, and for notifying WRA of any obvious violation of the CSIP requirements.

14.02. Grower and landowner requirements.

WRA shall require the growers and landowners to operate their irrigation systems and to use reclaimed water in accordance with the requirements of Water Reclamation Requirements Order No. 97-52, or as subsequently revised.

14.03. Reports to U.S. Department of the Interior, Bureau of Reclamation.

WRA shall prepare and submit all reports required by the Bureau of Reclamation for CSIP, and SRDF and PCA shall submit all such reports for SVRP, as a condition of any loans made by the Bureau of Reclamation. These reports shall include, but not be limited to, the annual excess lands report and any reports associated with use of reclaimed water for municipal or industrial purposes.

14.04. Permits and approvals.

WRA shall be responsible for obtaining and complying with all permits and approvals for the CSIP and SRDF that are necessary for PCA to perform its work under this Water Recycling Agreement.

14.05. Third-party damages.

WRA shall resolve and pay for all costs resulting from damages to CSIP and SRDF facilities caused by third parties and not covered by insurance maintained by either WRA or PCA. If such damages are caused by a grower who is being served by the CSIP, WRA shall determine whether the associated costs are to be recovered from the grower, or are to be absorbed by WRA, but in no case shall PCA have any cost or other liability with regard to repairing said damages, except that WRA may request PCA to make necessary repairs and reimburse PCA for all costs associated therewith. The budget for the CSIP does not include any amounts for making such repairs, and WRA shall fund the cost of such repairs outside of this Water Recycling Agreement.

14.06. Assessments.

WRA will prepare and submit, and resolve all questions and issues associated with billings for assessments levied in conjunction with SVRP, CSIP and SRDF.

14.07. Physical damage to crops.

Crop damage payments, except those attributed to PCA's failure to produce water which meets the water quality standards specified in Section 6.01 shall be reimbursable costs from the annual CSIP operating budget.

XV. DELIVERY OF WATER TO PCA.

15.01. PCA Demand.

PCA shall have the right to use, on an "as available" basis, amounts of secondary effluent or tertiary treated water to which WRA and/or MCWD are entitled, but do not use. WRA shall similarly have the right to use, on an "as available" basis, amounts which PCA and/or MCWD are entitled, but do not use.

15.02. PCA Monthly Demand Schedule.

PCA shall, no later than December 15 of each year, provide to WRA a schedule of tertiary treated water deliveries for the next calendar year. This schedule shall show the amount requested for each calendar month of the year. The demand may vary from month to month, provided the demand neither exceeds PCA's entitlement nor the seasonal limitations, except as otherwise specifically provided by this Water Recycling Agreement. PCA shall promptly inform WRA of any changes in the foregoing schedule.

15.03. PCA Payment.

1. PCA shall pay the Actual Costs of tertiary treated water it takes delivery of for its use, including portions of payments required pursuant to the Bureau Loan Contract and the

SWRCB Loan Contract. The "Actual Costs" consist of the sum of: 1) the proportional SVRP operation and maintenance, plus 2) Supplemental Well Pumping Costs, if any plus 3) the proportional cost of SVRP debt service for the SWRCB Loan Contract, plus 4) the proportional cost of principal and interest payments for the SVRP due pursuant to Article 9 (b) of the Bureau Loan Contract, including any proportional interest due pursuant to Article 9 (b) (2) of the Bureau Loan Contract as a result of deliveries of tertiary treated water to the PCA for its M&I use.

2. PCA payments will be determined and made consistent with Exhibit H and, to the extent applicable, Articles VI and VII of this Water Recycling Agreement.

15.04. No Modification of MCWD Contract Entitlement.

Nothing in this Water Recycling Agreement is intended to, nor shall it be interpreted to, expand, limit or otherwise modify MCWD's existing contractual rights, entitlements, and obligations pursuant to either of the Annexation Agreements.

XVI. GENERAL PROVISIONS.

16.01. Compliance with laws.

PCA will comply with all permit and licensing requirements applicable to the SVRP, CSIP and SRDF, and will operate the SVRP in accordance with all requirements of law and governmental regulations. Compliance with water quality requirements will be limited strictly to those set forth in Section 6.01, "Water quality."

16.02. Attorney's fees.

In the event it shall become necessary to commence or defend litigation for purposes of enforcing this Agreement or rights hereunder, the prevailing party shall be entitled to recover reasonable attorney's fees and costs.

16.03. Amendments.

No amendment or modification shall be made to this Water Recycling Agreement, except in writing, approved by the respective Boards and duly signed by both Parties.

16.04. Contract administrators.

- 1. WRA hereby designates its General Manager as its contract administrator for this Agreement. All matters concerning this Agreement which are within the responsibility of WRA shall be under the direction of or shall be submitted to the General Manager or such other WRA employee in the WRA as the General Manager may appoint. WRA may, in its sole discretion, change its designation of the contract administrator and shall promptly give written notice to PCA of any such change.
- 2. PCA hereby designates its General Manager as its contract administrator for this Agreement. All matters concerning this Agreement which are within the responsibility of PCA shall be under the direction of or shall be submitted to the General Manager or such other PCA employee in the PCA as the General Manager may appoint. PCA may, in its sole discretion, change its designation of the contract administrator and shall promptly give written notice to WRA of any such change.

16.05. Assignment.

Any assignment of this Water Recycling Agreement shall be void without the written consent of the

non-assigning party.

16.06. Negotiated Water Recycling Agreement.

This Water Recycling Agreement has been arrived at through negotiation between the parties. Neither Party is to be deemed the party which prepared this Water Recycling Agreement within the meaning of Civil Code Sec. 1654.

16.07. Time is of essence.

Time is of the essence of this Water Recycling Agreement.

16.08. Headings.

The article and paragraph headings are for convenience only and shall not be used to limit or interpret the terms of this Water Recycling Agreement.

16.09. Entire Water Recycling Agreement.

This written Water Recycling Agreement, together with all exhibits attached hereto and incorporated by reference, is the complete and exclusive statement of the mutual understanding of the parties, except to the extent that this Water Recycling Agreement expressly refers to or requires the preparation of additional agreements. Any such additional agreement shall be in writing. This Water Recycling Agreement supersedes and cancels all previous written and oral agreements and communications relating to the subject matter of the contract. The 1992 Agreement, Amendments 1, 2, and 3 dated May 30, 1994, February 16, 1998, and May 28, 2002, respectively, and the SRDF Agreement are hereby rescinded.

16.10. Notices.

All notices and demands required under this Water Recycling Agreement shall be deemed given by one party when delivered personally to the principal office of the other party; when faxed to the other party, to the fax number provided by the receiving party; or five days after the document is placed in the US mail, certified mail and return receipt requested, addressed to the other party as follows:

To PCA:

To WRA:

General Manager MRWPCA 5 Harris Court, Building D Monterey, CA 93940 Fax: (831) 372-6178

General Manager MCWRA P.O. Box 930 Salinas, CA 93902-0930 Fax: (831) 424-7935

16.11. Execution of documents.

The parties will execute all documents necessary to complete their performance under this Water Recycling Agreement.

16.12. Exhibits.

The following exhibits are attached to this Water Recycling Agreement:

EXHIBIT A: DIAGRAM OF PROJECT LOCATION AND RIGHTS OF WAY, ALSO SHOWING POINTS OF DELIVERY TO THE CSIP EXHIBIT B: OPERATION AND MAINTENANCE OF SVRP, CSIP, SRDF, AND NEW SOURCE WATER FACILITIES

EXHIBIT C:	SAMPLE WRA BASIC DEMAND SCHEDULE
	WASTEWATER RECLAMATION SYSTEM PROJECT DESCRIPTION AND
	MITIGATION MEASURES
EXHIBIT E:	CALCULATION OF PAYMENTS PURSUANT TO SECTION 17.08 OF THIS
	WATER RECYCLING AGREEMENT
EXHIBIT F:	OPERATION AND MAINTENANCE ESTIMATED TREATMENT COSTS
EXHIBIT G:	PROPOSED GWR PROJECT FACILITIES OVERVIEW FIGURE S-1
EXHIBIT H:	DEBT SERVICE OVERVIEW AND AMORTIZATION SCHEDULE
EXHIBIT I:	SCHEDULE OF AMORTIZED REPLACEMENT AND RENEWAL COSTS FOR
	NEW SOURCE WATER FACILITIES
EXHIBIT J:	SAMPLE COST ALLOCATION PLAN FOR INDIRECT COSTS

16.13. Severability.

If any provision of this Water Recycling Agreement is declared invalid or unenforceable by any court of competent jurisdiction, then such portion or provision shall be deemed to be severable, to the extent invalid or unenforceable, from this Water Recycling Agreement. Such declaration shall not affect the remainder of this Water Recycling Agreement, which shall remain in full force and effect, as though the invalid portion had never been included.

16.14. Waiver.

No waiver of any right or obligation of any of the parties shall be effective unless in a writing, specifying such waiver, executed by the party against whom such waiver is sought to be enforced. A waiver by any of the parties of any of its rights under this Water Recycling Agreement on any occasion shall not be a bar to the exercise of the same right on any subsequent occasion or of any other right at any time.

16.15. Conditions Precedent for New Source Water Facilities.

The portions of this Water Recycling Agreement applicable to the New Source Water Facilities (see Section I) shall not become effective until the following conditions are met:

- 1. Water Rights for the Blanco Drain and Reclamation Ditch are obtained from the California State Water Resources Control Board; and
- 2. A fully executed, and California Public Utilities Commission approved, Water Purchase Agreement, between PCA, Monterey Peninsula Water Management District, and California-American Water is approved by the California Public Utilities Commission and executed by the parties thereto; and
- 3. Written findings are made by the Regional Water Quality Control Board that utilization of the Blanco Drain dry weather flows as New Source Water meets all treatment requirements for the aforesaid dry weather flows; and
- 4. An independent third-party review of proposed capital and operating costs and preparation of an Engineer's Report is approved by the WRA Board of Directors and Board of Supervisors. The costs of the aforesaid third party review shall be shared equally between WRA and PCA; and
- 5. A successful assessment or Proposition 218 process for rates and charges related to the operation and maintenance of the New Source Water Facilities and proportional primary and secondary treatment charges; and
- 6. Inclusion of Salinas Pond Water Return Facilities as New Source Water Facilities requires execution of a separate agreement between the Parties.

16.16. Use of New Source Water.

If Conditions Precedent in Section 16.15 (1) and 16.15 (2) are met, but Conditions Precedent in Section 16.15 (3) and/or Section 16.15 (4) and/or Section 16.15 (5) are not met:

- 1. WRA will allow use by PCA of its water rights of 6,500 acre-ft entitlement from Application Nos. 32263A, 32263B, and 32263C. PCA will pay WRA the cost of obtaining and maintaining those water rights.
- 2. WRA will retain the right to utilize the Agricultural Wash Water component from the City of Salinas.
- 3. WRA and PCA will incorporate the provisions of this Section 16.16 in a separate agreement should Section 16.16 become operable.

16.17. Written Authorization.

For any action by PCA which requires written authorization from WRA, the written authorization shall be signed by WRA's General Manager, or the General Manager's written designee.

XVII. EXECUTION

In witness whereof, the parties execute this Water Recycling Agreement as follows:

Monterey Regional Water **Pollution Control Agency**

November 30. 201SDated:

Board Chair, Board of Directors

Approved as to form: ov. 30, 2015 Dated:

Counsel, Monterey Regional Water Pollution Control Agency

Monterey County Water Resources Agency

Dated:

Chair, Board of Supervisors

11116 Dated:

Deputy County Counsel, Monterey County Water Resources Agency

EXHIBIT A



EXHIBIT B

Operation and Maintenance of SVRP, CSIP, SRDF, and New Source Water Facilities

Operation and Maintenance of the SVRP

PCA agrees to perform the following minimum services to ensure the proper operation of the SVRP. Said services are subject to change by mutual written agreement.

- Operate mechanical equipment such as meters, pumps, filters and chemical feeders.
- Operate electrical and electronic equipment such as motors, controllers, automatic monitors, automatic samplers, chlorine residual analyzers, and standby power systems.
- Calibrate, maintain, service, repair, and replace various mechanical electrical, and electronic equipment such as chemical metering pumps and supporting chemical feed systems, automatic samplers, electrically and pneumatically operated valves, motors, flow meters, chemical mixers, and chlorine and sulfur dioxide equipment including evaporators, chlorine and sulfur dioxide feeders.
- Change one ton chlorine and sulfur dioxide tanks and abide by the PCA's safety and loss prevention program.
- Perform jar testing to determine and optimize chemical dosages and control chemical applications for various treatment processes.
- Inventory, order and store chemicals for the SVRP.
- Inventory, order and store spare parts and equipment for the SVRP.
- Maintain accurate and complete operational records and prepare monthly and any other reports as required by the Water Reclamation Requirements referred to in Section 5.01, paragraph (b) for the SVRP. Copies of Water Reclamation Requirements reports will be provided to WRA.
- Read meters, gauges, charts.
- Utilize computers and SCADA system to assist with the operation of the SVRP.
- Read and interpret SVRP construction schematics and plans.
- Maintain and monitor reclaimed water storage reservoir.
- Collect samples from throughout the various SVRP processes to ensure and optimize reclaimed water production.
- Perform laboratory analyses and studies as necessary to ensure compliance with Title 22 and water safety concerns.
- Abide by PCA's safety and loss prevention program policies and maintain a safe working environment.
- Perform regular preventive maintenance on the SVRP equipment.
- Perform general maintenance, housekeeping and grounds maintenance.
- Assist with, and provide support for, tours of the SVRP for members of the public.

Operation and maintenance of the CSIP

PCA agrees to perform the following minimum services to ensure the proper operation of the CSIP. Said services are subject to change by mutual written agreement.

Exhibit B Page 1 of 5 Perform normal operation and maintenance on the distribution well pumps, booster stations, pipelines, instrumentation, SCADA system and cathodic protection system. Perform repairs and major maintenance services as required due to wear and tear or failure of equipment. These will be performed as prescribed in the System Operations & Maintenance Manual prepared for the CSIP by WRA's design engineer, as well as the CSIP equipment manufacturers' manuals furnished in accordance with the CSIP construction contract documents, including:

- Inspect the pipelines, supplemental wells and booster stations.
- Inspect the pipelines for leaks.
- Inspect supplemental wells for general condition, log entries, fill pump bearing oilers, and perform general housekeeping.
- Inspect of booster stations for general condition, log entries, and perform general housekeeping.
- Inspect the turn-outs for serviceability.
- Inspect and calibrate conductivity meters.
- Read power and water meters at turnouts and supplemental wells once a month for records, and once a quarter at turnouts for billing purposes.
- Respond to after-hours problems on an on-call basis.
- Perform housekeeping of all assigned areas.
- Receive water orders and schedule water deliveries.
- Assess capability of the SVRP and the CSIP to supply water orders at the times and in the quantities requested.
- Notify growers as to whether each water order can or cannot be filled.
- Monitor turn-out flow rates to insure they conform with designed flow rate.
- Read water meters monthly.
- Monitor water quality within the distribution system and perform laboratory analyses and special studies as necessary to confirm the safety and public acceptance of the water.
- Maintain proper operation of air release and vacuum relief valves, and periodically exercise isolation valves.
- Maintain accurate and complete operational records and prepare reports as required by the recycled water user requirements referred to in Section 15.01 for the CSIP Copies of these reports will be provided to WRA.
- As necessary, replace motors, bearings and gaskets; repair valves and electrical or instrumentation equipment; and remove equipment from installed locations and transport to repair facilities.

Operation and maintenance of the SRDF

The SRDF is a seasonally operated facility. WRA shall notify PCA when SRDF seasonal operations are to begin and cease. As governed by project permits, raising the inflatable dam and pumping and chlorinating of river water may occur no sooner than April 1 of each year. Cessation of pumping and chlorinating river water shall occur no later than October 31 of each year. At the end of each operational season, the river water impoundment shall be filled to capacity and allowed to drain at 2 cfs through November 29 of each year unless directed otherwise by WRA. Depending on service area water demand, river water availability, quality, or other circumstances, raising of

Exhibit B Page 2 of 5 the inflatable dam and pumping of river water may begin after April 1, and cessation of pumping river water may occur before October 31.

The PCA agrees to perform the following services to ensure the proper operation of the SRDF (including the chlorination system): Perform repairs and maintenance services as required due to wear and tear or failure of equipment. These will be performed as prescribed in the Operations & Maintenance Manual prepared for the SRDF by WRA's design engineer, as well as equipment manufacturers' manuals furnished in accordance with Project construction contract documents, including chlorination system equipment manufacturer's manuals, and as described, but not limited to, below:

- Operate mechanical equipment such as inflatable dam gate and regulating weir, air compressors and compressed air system components; river intake screens, fish ladder regulating gates and associated hydraulic system; vertical turbine water pumps; hydraulic surge control system components; river water pipeline; pipeline standpipe and control valves; filter screen station, valves, backwash pumps and system components.
- Operate electrical and electronic equipment such as motors, controllers, automatic monitors, automatic samplers, chlorine residual analyzers, and standby power systems.
- Operate fish bypass system to conform to fishery agency bypass flow requirements. PCA shall periodically monitor the fish bypass system for flow verification. PCA shall not be responsible for determining fish bypass flow requirements, but shall be notified by WRA as to fish bypass flow requirements.
- Calibrate, maintain, service, repair, and replace various mechanical, electrical, and electronic equipment such as chemical metering pumps and supporting chemical feed systems, automatic samplers, electrically and pneumatically operated valves, motors, flow meters, chemical mixers, and chlorine equipment including evaporators and feeders.
- Prepare project components for operation prior to seasonal start up. This includes, but is not limited to, removing sediment from the river intake head wall area, fish ladder, and regulating weir, installing river intake screens, installing fish ladder grating, installing hand rails and safety equipment at the river head walls, initial operation of inflatable dam and hydraulic regulating gates, preparing hydraulic surge system for operation, opening and monitoring standpipe valve operation, preparing filter station components, preparing chlorination system components for operation, start up of air conditioning/ventilation systems, and installing debris boom.
- Close down project components after the operational season ends. This includes, but is not limited to, removing river intake screens, installing fish ladder cover plates, removing hand rails at the river head walls, lowering inflatable dam, closing hydraulic regulating gates, draining the river water pipeline, draining and decompressing hydraulic surge system, draining standpipe and closing valves, draining filter station components, draining chlorination system components, shut down of air conditioning/ventilation units, and removal of debris boom.
- Remove debris and sediment from any project component as needed to maintain adequate operation of project and prevent undue damage to project facilities.
- Regularly communicate forecast river water irrigation demand with WRA personnel. Such information will be used by WRA to assist determination of water releases from Nacimiento and/or San Antonio Reservoirs.
- Inspect and maintain painting and coatings to prevent atmospheric corrosion, and monitor and maintain buried facility corrosion protection equipment.
- Inspect or obtain proper inspection of pressure vessels.

- Perform testing as needed to determine and optimize chemical dosages and control chemical applications for river water treatment process.
- Inventory, order and store chemicals for the SRDF.
- Inventory, order and store spare parts and equipment for the SRDF.
- Maintain accurate and complete operational records and prepare monthly and any other reports as required by the WRA, or regulatory bodies.
- Read meters, gauges, and charts.
- Utilize computers and SCADA system to assist with the operation of the SRDF.
- Maintain and monitor reclaimed water storage reservoir.
- Collect samples from throughout the various SRDF processes to ensure and optimize river water production.
- Perform laboratory analyses and studies as necessary to ensure compliance with water safety concerns.
- Abide by PCA's safety and loss prevention program policies and maintain a safe working environment.
- Perform regular preventive maintenance on the SRDF equipment. Such maintenance may be performed in the off-season if determined to be more cost effective.
- Perform general maintenance, housekeeping and grounds maintenance of all assigned areas.
- Assist with, and provide support for, tours of the SRDF for members of the public.
- Inspect the pipelines for leaks.
- Assist WRA's grower liaison with the operation of the SRDF.
- Respond to after-hours problems on an on-call basis.
- Receive water orders and schedule water deliveries.
- Assess capability of the SRDF to supply water orders at the times and in the quantities requested.
- Notify growers as to whether water orders cannot be filled, and coordinate filling of the water orders as necessary.
- Monitor water quality within the delivery system and perform laboratory analyses and special studies as necessary to confirm the safety and public acceptance of the water.
- As necessary, replace motors, bearings and gaskets; repair valves and electrical or instrumentation equipment; and remove equipment from installed locations, transport to repair facilities and back to the project.
- PCA will be responsible for the care, maintenance, and use of additional equipment to be furnished by WRA for PCA's use in providing service under this agreement.
- Maintain and repair signage at SRDF site as needed. Signage will be supplied and installed by WRA.
- Implement the River Water Contaminant Response Plan as required by Plan criteria.

In addition to those services listed above, WPCA shall operate and maintain the SRDF river water chlorination system to conform to the following parameters:

- The chlorination system shall be capable of operation whenever the SRDF is operational, 24 hours per day, 7 days per week.
- The chlorination system shall be maintained to operate under normal flow capacity ranges from approximately 12 cubic feet per second (cfs) to 36 cfs. Maximum flow capacity is up to 48 cfs. Maximum flow capacity is intended only for short durations of significant water supply loss, such as when the SVRP is out of operation.
- A two-phase river water disinfection process including contact with free chlorine and

contact with combined chlorine via mixing with recycled water.

- River water contact with free chlorine for a minimum often (10) minutes prior to mixing with recycled water at 36 cfs flow.
- A free chlorine dosage capability up to 15 ppm.
- Promote river water and recycled water mixing prior to distribution.
- Prevent backflow of chlorinated water into the SRDF pipeline.
- Minimize effects of hydraulic surge pressures on the SRDF pipeline and pump station.
- Provide and maintain outlet for SRDF screen filter station backflush discharge water.
- Oversee modifications to the system by registered professional engineers according to accepted engineering standards and applicable building codes and regulations.
- Incorporate all appropriate safety features.
- River water sampling and testing features and locations.
- Maintain integrated river water disinfection system instrumentation and control features with the SRDF such that the SRDF operates in conjunction with the SVRP and CSIP under normal operating conditions.

Operation and maintenance of the New Source Water Facilities

PCA agrees to perform the following minimum services to ensure the proper operation of the new diversion structures. Said services are subject to change by mutual written agreement.

Perform normal operation and maintenance on the distribution well pumps, booster stations, pipelines, instrumentation, SCADA system and cathodic protection system. Perform repairs and major maintenance services as required due to wear and tear or failure of equipment. The service and repairs will be performed as prescribed in the System Operations & Maintenance Manual prepared for the facilities. These manuals will be similar to the equipment manufacturers' manuals furnished in accordance with the CSIP and SVRP construction contract documents. Some of the operation and maintenance items for the New Source Water Diversion Facilities include but not limited to:

- Calibrate, maintain, service, repair, and replace various mechanical, electrical, and electronic equipment
- As necessary, replace motors, bearings and gaskets; repair valves and electrical or instrumentation equipment; and remove equipment from installed locations, transport to repair facilities and back to the project.
- Read meters, gauges, charts.
- Utilize computers and SCADA system to assist with the operation of the diversion structures
- Abide by PCA's safety and loss prevention program policies and maintain a safe working environment.
- Perform regular preventive maintenance on the diversion equipment.
- Perform general maintenance, housekeeping and grounds maintenance
- Respond to after-hours problems on an on-call basis.

EXHIBIT C SAMPLE WRA BASIC DEMAND SCHEDULE FY 2015-2016 CSIP/SVRP/SRDF

DEMAND SCHEDULE

MONTH	PROJECTED RTP INFLUENT ACRE-FEET ¹	CSIP SUPPLEMENTAL WELLS PRODUCTION ACRE-FEET	SRDF RIVER PRODUCTION ACRE-FEET	SVRP + NEW SOURCE WATERS PRODUCTION ACRE-FEET	NEW SOURCE WATERS PRODUCTION ACRE-FEET	TOTAL DEMAND ACRE-FEET
July	1,976	1,355		2,045	245	3,400
August	1,959	802	-	2,198	348	3,000
September	1,864	197		2,003	303	2,200
October	1,930	20	-	1,268	300	1,288
November	1,767	70	-	786	200	856
December	1,783	100	-	_	-	100
January	1,601	250	-	-	-	250
February	1,491	150	_	806	150	956
March	1,620	130	_	1,670	170	1,800
April	1,858	450	-	1,950	250	2,400
Мау	1,946	1,210	-	2,040	240	3,250
June	1,894	1,324		1,926	226	3,250
TOTALS	21,689	6,058		16,692	2,432	22,750

<u>NOTES:</u> (1) Based on actuals November previous calendar year through October of current calendar year

Exhibit C Page 1 of 1
EXHIBIT D

WASTEWATER RECLAMATION SYSTEM

PROJECT DESCRIPTION

General description of the system

The following facilities constitute the Wastewater Reclamation System, under the contract with the MRWPCA:

1. A structure to divert the flow of secondary treated wastewater from the effluent pipeline of the existing regional treatment plant;

2. A pump station and pipeline to pump treated effluent to the tertiary treatment plant;

3. The 29.6 mgd tertiary treatment plant, consisting of coagulation, flocculation, filtration, and disinfection facilities;

4. Storage pond for tertiary treated water, with approximately 80 acre-feet of storage capacity;

5. Two pipelines connecting the reclamation facilities to the Castroville Seawater Intrusion Project (CSIP):

- a. A 54-inch diameter-main line from the Storage Pond, under the river, and connected to CSIP.
- b. A 27-inch diameter pipeline lateral under the Salinas River between portions of CSIP

Exhibit D Page 1 of 1

EXHIBIT E

CALCULATION OF PCA PAYMENTS PURSUANT TO SECTION 15.03 OF THIS WATER RECYCLING AGREEMENT

As required by Section 17.08 of this Water Recycling Agreement, PCA shall pay the Actual Costs of tertiary treated water to which it takes delivery for its use. The Actual Costs will be calculated from the cost components which comprise PCA's existing budget and actual spreadsheet model(s) for the annual SVRP and CSIP costs.

The four cost elements that compromise the Actual Costs will be computed as follows:

- (1) SVRP O&M Costs: The SVRP O&M budget will be calculated based on the Total Tertiary Treated Water Production needed to serve both the CSIP and M&I uses of tertiary treated water. The amount to be paid by PCA will be computed using the formula below,in which: A= Total Tertiary Treated Water Production, AFY
 - B =Tertiary Treated Water Production delivered to PCA for PCA use, AFY
 - C = Total O&M Expenditures from the SVRP O&M budget, excluding debt service
 - D =Amount to be paid by PCA

(2) <u>Supplemental Well Pumping Cost:</u> The amount to be paid by PCA will be computed as indicated in Section 2.G. of Amendment No. 3.

S = Amount to be paid by PCA

- (3) <u>SWRCB Loan Contract Debt Service:</u> The amount to be paid by PCA will be computed using the formula below, in which:
 - A= Total Tertiary Treated Water Production, AFY
 - B = Tertiary Treated Water Production delivered to PCA for its use, AFY
 - E = SVRP SWRCB Loan Debt Service
 - F = Amount to be paid by PCA

- (4) <u>Bureau Loan Contract Debt Service:</u> The amount to be paid by PCA will be computed using the formula below, in which:
 - A = Total Tertiary Treated Water Production, AFY
 - B =Tertiary Treated Water Production delivered to PCA for its use, AFY
 - G = SVRP Bureau Loan Debt Service, computed pursuant to Article 9 (b) (1) and 9 (c) of Bureau Loan Contract
 - H = Additional interest charged by the Bureau for the Bureau loan on the SVRP pursuant to Article 9 (b) (2) of Bureau Loan Contract
 - I = Amount to be paid by PCA

$$= \frac{B}{A} \times G + H$$

TOTAL PCA PAYMENTS= D + S + F + I

Exhibit E Page 1 of 1

EXHIBIT F

OPERATION AND MAINTENANCE ESTIMATED TREATMENT COSTS

These are the fees for primary and secondary treatment using the interruptible rate

The Equation for Special Users (using Ag Wash Water as example):

- Where K = Annual user charge for each user
 - V = Average daily flow in mgd (current estimate 3.332 mgd)
 - B = Average BOD concentration in mg/L (current 735 mg/L)
 - S = Average SS concentration in mg/L (current 282 mg/L)
 - C = Annual customer charge (currently \$2.75)
 - M = Annual maintenance charge (currently \$1,034.30)
 - I = Annual insurance (currently \$0)
 - L = Annual lab testing (currently \$0)
 - T = Annual testing (currently \$0)
 - R = Annual replacement funds (currently \$0 as covered by Article 3.01 5)

K = (71,757.72 * V) + (145.11 * V * B) + (149.99 * V * S) + C + M + I + L + T + R = \$736,388.58/year or \$197.32/AF

Table 1: Approved Rate Equat	ion Factors for FY15/16 and FY16/17:
------------------------------	--------------------------------------

Equation Factors	Sewage Rate FY15/16	Interruptible Rate FY15/16	Interruptible Rate FY16/17
Flow (\$/mgd)	\$666,000.00	\$71,757.72	\$74,125.73
BOD (\$/mgd*mg/L)	\$789.00	\$145.11	\$149.90
SS (\$/mgd*mg/L)	\$326.00	\$149.99	\$154.94
Approved by DCA Peer	d Ostahan D. 204	r	

Approved by PCA Board October 8, 2015

Table 2: Those rate equation factors result in the following interruptible rates:

Water Type	FY15/16	FY16/17		
	Rate (\$/AF)	Rate (\$/AF)		
Ag Wash Water	\$197.32	\$203.83		
Blanco Drain	\$73.90	\$76.34		
Reclamation Ditch	\$76.92	\$79.46		
Pond Water (Ag Wash	\$83.20	405 OF		
and Storm)	ΦΟ Ο.20	\$85.95		
Tembladero Slough	\$91.02	\$94.02		
Salinas Storm Water	\$68.52	\$70.78		
Lake El Estero	\$68.32	\$70.57		

The above rates are based on flow (V), biochemical oxygen demand (B), and total suspended solids (S) of the New Source Waters. Changes in the water characteristics will change the fee. The Board will consider revising the fees as part of the FY17/18 budget process.

Exhibit F Page 1 of 1



EXHIBIT H

Debt Service & Amortization Schedule

Salinas Valley Reclamation Plant (SVRP) <u>United States Bureau of Reclamation Loan</u> Loan Number:	
Contract Number: 5-07-20-W1284	
Original Principal	\$ 20,444,141.34
Current Principal (as of 6/30/14)	\$ 14,694,226.56
Term	35 years (2003 – 2037)
Rate	Variable 2.27%
Principal Payment	\$ 638,879.42
Interest	\$ 344,901.42 (variable)
IDC	\$ 183,000.00 (variable)
Ending Principal (as of 6/30/15)	\$ 14,045,923.94
Total Payment for 2015:	\$ 1,180,169.88 (actual)

Notes:

Annual payment on this loan is due on or around April, 11^{th.} Invoice received by PCA from USDOI Bureau of reclamation. PCA invoices WRA for loan payment. Loan paid out of fund 132. Full amount is paid out of object 7201. Emergency Reserve Fund under Section 20 of the USBR Contract stipulates that annual deposits of \$13,000 will be made until the account reaches \$130,000 which is where is currently resides. Interest is still payable annually to the USBR. **The current Emergency Reserve amount is \$134,456.24**.

State Water Resources Control Board	
Loan Number: C-06-4017-210	
Contract Number: 5-804-550-0	
Original Principal	\$ 9,319,708.22
Current Principal (as of 6/30/14)	\$ 2,342,999.55
Term	20 years (1999 – 2018)
Rate	3.10%
Principal Payment	\$ 559,446.56
Interest	\$ 72,643.61
IDC	None
Ending Principal (6/30/15)	\$ 1,783,552.99
Total Payment for 2015:	\$ 632,090.19 (actual)
Notes:	
• • • • • • • • •	4

Annual payment on this loan is due on or around March, 6^{th.} Invoice received by PCA from SWRCB. PCA invoices WRA for loan payment. Loan paid out of fund 132. Full amount is paid out of object 7201. Article 8 of this contract requires a Wastewater Reserve Fund. The current amount in this reserve is \$120,000 held by MRWPCA and \$129,160.31 held by MCWRA.

Exhibit H Page 1 of 5

New Source Water Facilities

State Water Resources Control Board Original Principal Term Rate Estimated Principal Payment Estimated Interest Payment Estimated Total Annual Debt Service

\$3.9 Million (estimated) 30 Years 1% (estimated) \$112,117 (variable) \$39,000 (variable) \$151,117

Notes: Invoice projected to be received by PCA from the SWRCB. PCA will invoice WRA for the loan payment.

Exhibit H Page 2 of 5

Salinas Valley Reclamation Plant US Bureau of Reclamation Loan

			Principal	Interest on	Total	Total	Loan	interest
		Year	Payment	Loan	IDC Pmt	Payment	Balance	Rate
	2003	1	0	375,218.99	141,384.78	516,603.77	20 444 141 24	1 0250/
	2004	2	0	375,218.99		516,603.76	20,444,141.34	1.835%
	2005	3	0	-	133,124.46		20,444,141.34	1.835%
	2006	4	638,879.42	403,746.24		486,363.45	20,444,141.34	1.728%
	2007	5	638,879.42	419,520.01	163,203.77	1,194,784.68	19,805,261.92	1.975%
	2008	6	638,879.42		163,205.77	1,221,603.20	19,166,382.50	2.118%
	2009	7	638,879.42	417,883.20	-	1,214,217.07	18,527,503.08	2.141%
	2005	8	638,879.42	417,883.20	173,778.52	1,230,541.14	17,888,623.66	2.255%
2010		st Corre		2,455.22	-	1,216,131.37	17,249,744.24	2.255%
2010	2011	9			1,057.48	1 222 020 10	46 640 064 00	
	2011	9 10	638,879.42	401,295.89	179,242.15	1,222,930.16	16,610,864.82	2.326%
	2012	10	638,879.42	389,888.83	180,828.36	1,209,596.61	15,971,985.40	2.347%
	2013		638,879.42	374,893.10		1,194,600.88	15,333,105.98	2.347%
		12	638,879.42	359,897.13	180,828.36	1,179,604.91	14,694,226.56	2.347% Est
	2015	13	638,879.42	344,901.42	183,000.00	1,166,780.84	14,055,347.14	2.347% Est
	2016	14	638,879.42	329,905.70	183,000.00	1,151,785.12	13,416,467.72	2.347% Est
	2017	15	638,879.42	314,909.99	183,000.00	1,136,789.41	12,777,588.30	2.347% Est
	2018	16	638,879.42	299,914.27	183,000.00	1,121,793.69	12,138,708.88	2.347% Est
	2019	17	638,879.42	284,918.56	•	1,106,797.98	11,499,829.46	2.347% Est
	2020	18	638,879.42	269,922.85	183,000.00	1,091,802.27	10,860,950.04	2.347% Est
	2021	19	638,879.42	254,927.13	183,000.00	1,076,806.55	10,222,070.62	2.347% Est
	2022	20	638,879.42	239,931.42	183,000.00	1,061,810.84	9,583,191.20	2.347% Est
	2023	21	638,879.42	224,935.71	183,000.00	1,046,815.13	8,944,311.78	2.347% Est
	2024	22	638,879.42	209,939.99	183,000.00	1,031,819.41	8,305,432.36	2.347% Est
	2025	23	638,879.42	194,944.28	183,000.00	1,016,823.70	7,666,552.94	2.347% Est
	2026	24	638,879.42	179,948.56	183,000.00	1,001,827.98	7,027,673.52	2.347% Est
	2027	25	638,879.42	164,952.85	183,000.00	986,832.27	6,388,794.10	2.347% Est
	2028	26	638,879.42	149,957.14	183,000.00	971,836.56	5,749,914.68	2.347% Est
	2029	27	638,879.42		183,000.00	956,840.84	5,111,035.26	2.347% Est
	2030	28	638,879.42	119,965.71	183,000.00	941,845.13	4,472,155.84	2.347% Est
	2031	29	638,879.42	104,969.99	183,000.00	926,849.41	3,833,276.42	2.347% Est
	2032	30	638,879.42	89,974.28	183,000.00	911,853.70	3,194,397.00	2.347% Est
	2033	31	638,879.42	74,978.57	183,000.00	896,857.99	2,555,517.58	2.347% Est
	2034	32	638,879.42	59,982.85	183,000.00	881,862.27	1,916,638.16	2.347% Est
	2035	33	638,879.42	44,987.14	183,000.00	866,866.56	1,277,758.74	2.347% Est
	2036	34	638,879.42	29,991.43	183,000.00	851,870.85	638,879.32	2.347% Est
	2037	35	638,879.32	14,995.71	183,000.00	836,875.03	0	2.347% Est

Salinas Valley Reclamation Plant SWRCB - State Revolving Fund Loan

		Pri	incipal	Interest on	Total	Loan
١	fear	Pa	yment	Loan	Payment	t Balance
3/31/1998 Bal	lance					9,319,708.22
3/31/1999	1		0	632,672.76	632,672.7	
3/31/2000	2	626	,019.10	6,653.66	632,672.7	
3/31/2001	3	365	,193.60	267,479.16	632,672.7	
3/31/2002	4	376	,514.61	256,158.15	632,672.7	76 7,951,980.91
6/30/2002 Ad	j	72	,591.27	-72,591.27		0 7,879,389.64
3/31/2003	5	387	,829.11	244,261.08	632,090.1	.9 7,491,560.53
3/31/2004	6	399	,851.81	232,238.38	632,090.1	.9 7,091,708.72
3/31/2005	7	412	,247.22	219,842.97	632,090.1	.9 6,679,461.50
3/31/2006	8	425	,026.88	207,063.31	632,090.1	.9 6,254,434.62
3/31/2007	9	438	,202.72	193,887.47	632,090.1	.9 5,816,231.90
3/31/2008	10	452	,279.63	170,810.56	623,090.1	.9 5,363,952.27
3/31/2009	11	465	,807.67	166,282.52	632,090.1	.9 4,898,144.60
3/31/2010	12	480	,247.71	151,842.48	632,090.1	.9 4,417,896.89
3/31/2011	13	495	,135.39	136,954.80	632,090.1	.9 3,922,761.50
3/31/2012	14	510	,816.84	121,273.35	632,090.1	.9 3,411,944.66
3/31/2013	15	526	,319.91	105,770.28	632,090.1	.9 2,885,624.75
3/31/2014	16	542	,635.82	89,454.37	632,090.1	.9 2,342,988.93
3/31/2015	17	559	,457.53	72,632.66	632,090.1	.9 1,783,531.40
3/31/2016	18	576	,951.78	55,138.41	632,090.1	.9 1,206,579.62
3/31/2017	19	594	,686.22	37,403.97	632,090.1	9 611,893.40
3/31/2018	20	611	,893.40	18,968.70	630,862.1	.0 0

+

New Source Water Facilities - Amortization Schedule



	Loan summary
Scheduled payment	
Scheduled number of payments	
Actual number of payments	30

	Beginning Balance	Sche	eduled Payment	Тс	otal Payment		Principal		Interest		Ending Balance
\$	3,900,000.00	\$	151,117.64	\$	151,117.64	1	112,117.64	\$	39,000.00	\$	3,787,882.3
\$	3,787,882.36	\$	151,117.64	\$	151,117.64	\$	113,238.82	\$	37,878.82	ŝ	3,674,643.5
\$	3,674,643.54	\$	151,117.64	\$	151,117.64	\$	114,371.21	\$	36,746.44	Ś	3,560,272.3
\$ \$ \$	3,560,272.33	\$	151,117.64	\$	151,117.64	\$	115,514.92	\$	35,602.72	Ś	3,444,757.4
	3,444,757.42	\$	151,117.64	\$	151,117.64	\$	116,670.07	\$	34,447.57	Ś	3,328,087.3
\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	3,328,087.35	\$	151,1 <mark>17.6</mark> 4	\$	151,117.64	\$	117,836.77	\$	33,280.87	ŝ	3,210,250.5
\$	3,210,250.58	\$	151,1 <u>17.6</u> 4	\$	151,117.64	\$	119,015.14	\$	32,102.51	Ś	3,091,235.4
\$	3,091,235.44	\$	151,117.64	\$	151,117.64	\$	120,205.29	\$	30,912.35	ŝ	2,971,030.1
\$	2,971,030.15	\$	151,117.64	\$	151,117.64	\$	121,407.34	\$	29,710.30	\$	2,849,622.8
\$	2,849,622.82	\$	151,117.64	\$	151,117.64	\$	122,621.41	Ś	28,496.23	\$	2,727,001.4
\$	2,727,001.40	\$	151,117.64	\$	151,117.64	\$	123,847.63	\$	27,270.01	ŝ	2,603,153.7
\$	2,603,153.78	\$	151,117.64	\$	151,117.64	\$	125,086.10	\$	26,031.54	ŝ	2,478,067.6
\$	2,478,067.67	\$	151,117.64	\$	151,117.64	\$	126,336.96	\$	24,780.68	Ś	2,351,730,7
\$	2,351, 730.71	\$	151,117.64	\$	151,117.64	\$	127,600.33	\$	23,517.31	ŝ	2,224,130.3
\$	2,224,130.37	\$	151,117.64	\$	151,117.64	Ś	128,875.34	\$	22,241.30	Ś	2,095,254.0
\$	2,095,254.04	\$	151,117.64	\$	151,117.64	\$	130,165.10	\$	20,952.54	ŝ	1,965,088.9
\$	1,965,088.94	\$	151,117.64	\$	151,117.64	\$	131,466.75	\$	19,650.89	ŝ	1,833,622.1
\$	1,833,622.18	\$	151,117.64	\$	151,117.64	\$	132,781.42	\$	18,336.22	\$	1,700,840.7
\$ \$ \$ \$ \$	1,700,840.76	\$	151,117.64	\$	151,117.64	\$	134,109.23	\$	17,008.41	\$	1,566,731.5
\$	1,566,731.53	\$	151,117.64	\$	151,117.64	\$	135,450.33	ŝ	15,667.32	\$	1,431,281.2
\$	1,431,281.20	\$	151,117.64	\$	151,117.64	\$	136,804.83	ŝ	14,312.81	\$	1,294,476.3
\$	1,294,476.37	\$	151,117.64	\$	151,117.64	Ś	138,172.88	\$	12,944.76	\$	1,156,303.5
\$	1,156,303.50	\$	151,117.64	\$	151,117.64	\$	139,554.61	Ś	11,563.03	ŝ	1,016,748.8
\$	1,016,748.89	\$	151,117.64	\$	151,117.64	\$	140,950.15	\$	10,167.49	\$	875,798.7
\$	875,798.74	\$	151,117.64	\$	151,117.64	\$	142,359.65	\$	8,757.99	ŝ	733,439.0
\$	733,439.08	\$	151,117.64	\$	151,117.64	\$	143,783.25	\$	7,334.39	\$	589,655.8
\$ \$ \$ \$ \$ \$ \$	589,655.83	\$	151,117.64	\$	151,117.64	\$	145,221.08	\$	5,896.56	\$	444,434.7
	444,434.75	\$	151,117.64	\$	151,117.64	\$	146,673.29	\$	4,444.35	Ś	297,761.4
\$	297,761.45	\$	151,117.64	\$	151,117.64	\$	148,140.03	\$	2,977.61	\$	149,621,4
\$	149,621.43	Ś		Ś	149,621.43	ċ	148,125.21	Ś	1,496.21	\$	1-10,021.

<u>EXHIBIT I</u>

Schedule of Amortized Replacement and Renewal Costs New Source Water Facilities

DRAFT

Estimated Original Cost - Equipment Requiring Replacement		ortional Share 4.9%	١	WRA Proportional Share 45.1%	Т	otal Costs
Electrical Equipment & Sluice Gates (part of Construction Cost, above) Instrumentation Equipment (part of Construction Cost, above) Pumps & Motors	\$ \$ \$ \$	124,455 16,177 252,063 392,695	; ; ; ; ; ; ; ; ; ; ; ; ;	13,289	\$	226,695 29,466 459,130 715,291
Estimated Annual Proportional Contribution	Estim	ated Life	F	Replacement Frequency Thirty Year Term		nated Annual onal Amount (1)
Electrical Equipment & Sluice Gates (part of Construction Cost, above)		30		1		38,747
Instrumentation Equipment (part of Construction Cost, above)		15		2		4,448
Pumps & Motors		20		1.5		68,263
Estimated Annual Proportional Contribution					\$	111,458
				PCA Share	\$	61,191
(1) Annual inflation factor for replacement cost of equipment		5.60%		WRA Share	\$	50,268

(Based on Cal-American Water Company factor)

Exhibit I Page 1 of 1

EXHIBIT J





COST ALLOCATION PLAN OVERVIEW

Exhibit J Page 1 of 7

MRWPCA COST ALLOCATION PLAN

TABLE OF CONTENTS

Section 1.	Purpose1
Section 2.	Support Service Department Narratives3
Section 3.	Navigating the Cost Plan
Exhibits	
Α.	Cost Allocation Plan Detail Exhibit A

Exhibit J Page 2 of 7

PURPOSE

A Cost Allocation Plan is an analytical tool through which administrative or indirect services costs are apportioned to individual departments or cost centers within an organization. Allocated costs are typically termed "agency-wide overhead." Most overhead/indirect service costs are those expenditures that provide support services or oversight to each department, and typically include legislative, managerial, financial, administrative, legal, personnel, technology, building and equipment use, and risk management activities. Allocated costs may also include services within non-administrative/support services departments that provide targeted benefits to specific departments. A general example of this would be specific skill sets from one department that actually exist to provide support to a department outside its own.

A brief analysis of Monterey Regional Water Pollution Control's Agency's (MRWPCA) process for allocation costs are provided below:

- Each department within MRWPCA's budget represents a unique function. Each department can be classified as either an administrative or operational unit. Administrative departments include Administration, Finance, Human Resources, Information Technology, Regional Treatment Plant Administration, and Safety. These departments provide support for the operational units. Costs associated with administrative departments are allocated to departments as either overhead or indirect services.
- 2. A review of line-item expenditures are conducted within each department to determine whether there are any significant costs which should be excluded or included as overhead. Examples of exclusions include one-time expenditures for capital outlay, transfers between funds, and debt service. The remaining "allocable" expenditures are considered costs for distribution to the operational departments.
- 3. MRWPCA assigns an allocation factor as the basis for distributing overhead/indirect service costs to departments or cost centers receiving support. Allocation factors include employee count, percentage of operating budget, and number of invoices. The factors represent approximately how much time each administrative position/department spends time supporting operational departments or programs. Ultimately, the allocation factor represents the actual or estimated workload of the allocated function and a reasonable method of apportioning benefit for the services received.
- 4. MRWPCA estimates the cost allocations at the beginning of the budget year based on the number of employees, percentage of budget, and the three year average number of invoices associated with each operational unit. Each administrative cost center is distributed across departments using the selected allocation factor for each cost center. This results in a total share of overhead/indirect service costs for each operational department / cost center. All costs applied throughout the year are based on actual operational department expenses incurred.
- 5. MRWPCA has established the Cost Allocation Plan included in this exhibit for the purpose of allocating overhead to shared and contracted services; and partner agencies. This cost

Exhibit J Page 3 of 7 allocation plan was not established to be consistent with *Title 2*, *Code of Federal Regulations*, *Part 225*, *Cost Principles for State*, *Local*, *and Indian Tribal Governments (formerly known as OMB A-87). The* primary goal of the Title 2 Plan is typically to obtain reimbursement of overhead costs associated with State and Federal grants.

The MRWPCA Cost Allocation Plan has been reviewed by an independent consultant, NBS, and has been determined to be reasonable for this specified use. The MRWPCA Cost Allocation Plan may be replaced at any time with a Plan that is compliant with Title 2, CFR, Part 225 for the purpose of obtaining reimbursement of overhead costs associated with State and Federal Grants.

Data Sources

MRWPCA referenced the following data to support the development of the initial Cost Allocation Plan:

- Budgeted expenditures for Fiscal Year 2015/16
- Listing of all Agency positions by Department
- Number of invoices and purchase orders processed by each fund and department for Fiscal Years: 2012/13, 2013/14, and 2014/15

SUPPORT SERVICE DEPARTMENT NARRATIVES

The full cost allocation plan allocates six administrative support departments to various operational department cost centers, based on a variety of factors. The plan identifies total allocable costs at approximately \$3.37 million from the following administrative units for distribution to the operating functions:

Budget Account	Allocable Department Budget	Total Allocable Expense
01-010	Administration	\$1,106,677
01-013	Finance	916,737
01-011	Human Resources	383,624
01-015	Information Technology	242,449
01-030	RTP Administration	415,631
01-016	Safety	305,165
	Total Administrative Allocations	\$3,370,283

The following paragraphs provide a brief description of each Administrative service department, along with a general explanation of how administrative costs are allocated to each functional department within the Plan.

Administration (01-010)

The Administration Department is responsible for all aspects of administration, operation, and planning activities of the Agency staff. The General Manager and the Executive Assistant/Clerk to the Board's salaries are included in this budget. The Administration Department is responsible for assisting the Board with policy development, implementation of strategic goals, and providing Agency leadership and management. Costs include legal counsel, insurance, contract services, and utility costs for the Harris Court administrative offices. While the Stormwater Program Manager's salaries are included in the FY 15/16 personnel expenses, they are directly charged to

Exhibit J Page 4 of 7 contracting entities and removed from the cost allocation. Insurance costs paid directly to the Monterey County Water Resources Agency for liability coverage associated with the Salinas Valley Reclamation Project, Castroville Seaside Intrusion Project, and the Salinas River Diversion Facility operations have also been excluded from the allocation.

The Administration Department budget, net of excludable line items, is charged to operational departments based on their percentage of the operational budgets.

Finance (01-013)

The Finance Department provides Agency-wide fiscal support including preparation of the Comprehensive Annual Financial Report, monthly financial reports; and the development and monitoring of the Agency's budget. The professional level Finance staff provide fiscal analyses and guidance to the General Manager, Board of Directors, partner agencies, and departments. Finance support staff provide accounts payable, payroll processing, procurement, project accounting, general ledger maintenance, and banking services.

Costs associated with Finance staff are allocated based on the following factors:

- Chief Financial Officer and Accountant's salaries; along with a proportional share of the Finance Department expenses are allocated based on the operational units percentage of the budget
- Accounting Technician, Procurement Specialist, and Administrative Support Specialist salaries; along with a proportional share of Finance Department expenses are allocated based on a three-year average of invoices processed by the operational units
- Accounting Specialist Payroll, along with a proportional share of Finance Department expenses are allocated based on the percentage of full-time equivalent positions in each operating department

Human Resources (01-011)

The Human Resources Department supports the Agency by providing a wide range of personnel services such as benefits administration (health, dental, vision, employee assistance program), workers compensation, coordinating recruitment and selection; classification and compensation programs, as well as the administration of the Agency's personnel files, training records, medical files, and employee evaluations. The Department also provider employer/employee relations administrative services and contract negotiations, which includes employee and supervisory counseling on personnel policies, procedures, and practices; and providing assistance in handling of employee grievances.

 Costs associated with the Human Resources Department are allocated based on the percentage of full-time equivalent positions in each operating department

Information Systems (001-015)

The Information Systems Department is responsible for developing, maintaining, and facilitating the implementation of a sound and integrated IT architecture. This includes the installation, integration, evaluation, and maintenance of the Agency's local and wide area networks,

telephone, and audio video systems; while serving as a technical resource to staff on technology needs. The Information Systems Department defines, delivers, and supports strategic plans for implementing information technologies.

 The costs of the Information Systems budget are allocated based on the percentage of full-time equivalent positions in each operating department

RTP Administration (01-030)

The RTP Administration Department is a centralized support service team located at the Regional Treatment Plant Facility. The Assistant General Manager provides leadership to all departments located at the Regional Treatment Plant and works closely with the General Manager to accompany Agency goals. Costs included within this budget unit primarily consist of salary and benefits associated with the Assistant General Manager, Administrative Support Specialist, and office expenses.

All costs associated with the RTP Administration are allocated based on the following:

- General Manager salary and proportional share of expenses based on the operational department's percentage of the budget
- Administrative Support Specialist salary and proportional share of expenses based on the average percentage of budget and full-time equivalency positions within each operational department.

Safety (01-016)

The Safety Department provides service to all Agency employees though training and assistance in implementing safe work practices. The Safety Officer works to implement programs which meet and exceed Federal, State, and Regional requirements for work safety, contractor safety; and materials safety at all Agency facilities. The Safety Budget includes costs associated with Agency-wide safety training, protective equipment, and the safety shoe reimbursement program.

 Costs associated with the Safety Department are allocated based on the percentage of full-time equivalent positions in each operating department

Additional Notes

Administrative costs that are charged directly to specific functional areas or capital projects are not allocated as part of this cost allocation plan.

NAVIGATING THE COST ALLOCATION PLAN

An overview of the MRWPCA Cost Allocation Plan is listed below and attached to this report.

• Agency-Wide Chart – Illustrates the departments identified as administrative cost centers and departments identified as operational cost centers. The allocation percentage of each administrative cost center is listed below each administrative cost function.

Exhibit J Page 6 of 7 MRWPCA Cost Allocation Plan Overview





Board Report

Legistar File Number: WRAPA 21-022

June 04, 2021

Current Status: Draft Matter Type: WRA P&A Item

Receive an update on personnel activities





Introduced: 5/27/2021

Version: 1



Item No.3

Board of Supervisors Chambers

168 W. Alisal St., 1st Floor Salinas, CA 93901

Monterey County

Board Report

Legistar File Number: WRAPA 21-023

Introduced: 5/27/2021

Version: 1

Receive an update on the Training and Professional Growth Plan





168 W. Alisal St., 1st Floor Salinas, CA 93901 June 04, 2021

Current Status: Draft Matter Type: WRA P&A Item



Board of Supervisors Chambers

Monterey County

Board Report

Legistar File Number: WRAPA 21-024

June 04, 2021

Current Status: Draft Matter Type: WRA P&A Item

Receive an update on real property and Agency land appraisals



Introduced: 5/27/2021

Version: 1



Board of Supervisors Chambers

168 W. Alisal St., 1st Floor Salinas, CA 93901

Project List: Land Use and Leases (updated 5/25/2021)

Project	Category	Notes	Assigned To	Priority [High (3), Medium (2), Low (1)]
Fee Schedule and Forms for land use and lease requests	In Progress	Adopt a fee schedule to allow reimbursement for land use and lease related requests. Standardize approval procedures for all requests for access, easements, etc with forms and guidelines. Fees for services will be estimated and added to Article XI Agency Fees. Forms to be made available for public.	Jennifer Bodensteiner	3
Boat Dock Program Enhancements	In Progress	Increase inspections and processing violation against nuisance docks around the reservoir, below the highwater level of 800', and on Agency lands; Establish a new project/customer time and billing system for all docks and log-booms at Nacimiento Reservoir; Coordinate with ITD for 2021 secure online payment portal	Jennifer Bodensteiner	3
MCWRA vs Dietel	In Progress	Complete conditions and terms of 2013 Judgment including appraisal, sales agreement for land and roadway easement, fencing provision, and lot line adjustment	Jess Barreras	3
San Antonio and Nacimiento Lease with County of Monterey and MOU with MC Parks	Under Review	Parks and MCWRA to work out all details by November 2021. This goes with request by MC Parks for Prop 68 funds and new improvements for recreation at SA. Standing meetings between MCWRA, and Parks occur the first friday of every month to complete the leases.	Jennifer Bodensteiner	3
Heritage Ranch (on Agency waterfront lands) Long-Term Lease Agreement (\$?)	Pending Approval	Consider long-term access to allow continued use of Agency Agency property. Under a current, one-year annually renewal lease which expires 03/31/2020. Scope of work includes appraisal and agreement. Appraisal titled "The Point" at Nacimiento Lake - Market Rent Study 25-acre, by The Dore Group, dated 09/30/2020 values the rental at \$51,750 annually. The BOD has directed the GM to proceed in negotations with the HROA.	Jessell Fenley	3
Potential new grazing leases (south of Nacimiento Resort and north of HROA)	In Progress	Consider new grazing leases on Agency property at Naci Reservoir. Benefit would be fire suppression and increased revenue. Approved in concept by the BOS in late 2019. Neighboring parcel owner is interested in Agency parcels and is seeking to negotiate rental price / improvement credit. Scope of work, proper surveying, and consultation with MC PW, Parks, and Facilities is necessary to move forward with final package.	Jennifer Bodensteiner	3
Potential grazing lease additions from excess recreational lands (600 acres in San Antonio Parks back to MCWRA)	In Progress	Consider attaining excess lands back from MC PW, Parks, and Facilities for incorporation in new or existing grazing leases at SA Reservoir. Benefit would be fire suppression, increased revenue, and reduced excess recreational acreage. Final maps for the area are under production and external review.	Jennifer Bodensteiner	3
Tri-Counties (Cullen Site)	In Progress	Consider access request to trade or sell Agency lands to resolve property boundary issue resulting in cabin on Agency property. Old survey vs. new survey results in difference in lot line placement and cabin error. Research the Dietel settlement for ideas on way forward. They need letter from Agency saying we approve in concept (see Access Policy).	Jennifer Bodensteiner	3

Pinball Tower lease proposal (Wendy Hughes- Not Started Degolia) Proposal received dated Jan 19, 2021 to resolve antenna system on a property owned by Ms. Hughes-Degolia. Five options presented to move forward. Option preferred by requester is a land swap with MCWRA. Investigation needs to be completed to better understand issues and request.



3

Project	Category	Notes	Assigned To	Priority [High (3), Medium (2) Low (1)]
Trespass issue on the firebreak road north of Heritage Ranch	In Progress	Repeat occurances of hikers and recreational users trespassing on Agency property. The property is between Nacimiento Resorts and the Hertitage Ranch subdivision. Agency is proposing to remedy by lockouting access for the hikers with simple modifications to the walk-through gate and fix the adjacent fencing.	Tom Shepherd	3
Tri-Counties (on Agency waterfront lands "The Flats") Long-Term Lease Agreement (\$?)	Not Started	Consider long-term access request from owner to allow continue use of lands and an established launch ramp on Agency property. Ramp has been in use for appx 40 years and they say were granted prior approval from Agency. Agreement will be to Access the flats; to have sole use of the launch ramp; and, to place private boat docks on agency property. Lease 3B amendment completed. Scope of work includes appraisal and agreement approval paid at Tri-Counties expense.	Jennifer Bodensteiner	3
Town Creek Assocation (on Agency waterfront lands) Long-Term Lease Agreement (\$?)	Not Started	Consider long-term access to allow continued historic use of Agency lands and construction/installation of a permanent launch ramp. Request to construct, operate, and maintain concrete launch ramps on Agency parcel. Lease 3B amendment completed in short-term lease negotiations. Scope of work includes design plans, appraisal and agreement paid at Town Creek's expense.	Jennifer Bodensteiner	2
Ray Franscioni (well)	Not Started	Talk to B. Buche? Is this an issue.	undetermined	2
Ray Franscioni (pipeline)	Not Started	Talk to B. Buche? Is this an issue.	undetermined	2
Telecommunication towers in SLO County	Not Started	2-40 acre Agency parcel adjacent to Hearst ranch. SLO County interested in access for communication purposes.	Jennifer Bodensteiner	2
Haag	Under Review	Request to convey road easement to private property parcel at San Antonio Reservoir.	Jess Barreras	2
Cameras at grazing leases	Not Started	Install game type cameras on grazing leases and vacant Agency lands. Purpose to deter trespassing.	undetermined	2
Oak Shores II (launch ramp)	In Progress	Request received in Dec 2020 to process Agency authorization to construct new launch ramp at Nacimiento Reservoir. Mass grading was completed 10 years ago. The developer has hired engineer to get approvals through final building permit approvals.	Jennifer Bodensteiner	2
Cal Shasta (on Agency waterfront lands) Potential new recreational lease on the "Cross Parcel" (\$?)	Not Started	Cal Shasta Club is using Agency land without authorization or agreement. Contact needs to be made with Cal Shasta Club and evaluation of existing and potential use of Agency lands.	Jennifer Bodensteiner	1
Jaureguy	Not Started	New grazing lease. Request for sheep grazing on Agency property (? Acres) near Lease 2. He was recommended to submit a proposal to Agency. He sent a check in and it was returned.	undetermined	1
Jaureguy (Lease 2)	Not Started	Agency parcel Lease 2 has a trash pile / old dump / garbage pit that needs removal. Tom will provide a photo of the garbage. Are people adding to it??	Jessell Fenley	1
Telecommunication towers in SLO County	Not Started	2-40 acre Agency parcel adjacent to Hearst ranch. SLO County interested in access for communication purposes.	undetermined	1
Nacimiento Water Company (repair and replace meters; rate increase)	Not Started	Naci water company purchases water from Agency. In Agreement certain meter is specified and current meter does not meet standard.	Jennifer Bodensteiner	1

Project	Category	Notes	Assigned To	Priority [High (3), Medium (2), Low (1)]
Pleyto Cemetery Road drainage damage and 1,000' fence project	Not Started	Repair damaged road to Vanness easement from storm damage on county road. MC RMA-PW were working on a FEMA grant to cover the cost. Tom will get an update from David Gilis. Fence needed to keep public off and vehicles off of agency property. Fence type bob-wire, cattle type boundary fence. Estimate \$5K.	undetermined	1
US Army and Tank Road at SA Reservoir	Not Started	Existing Access and easement Agreement between parties. Re- evaluate who should be on Tank Road? Should gates be closed? Comm wire removal (80% completed)? Boundary fence in reservoir bottom? Wildlife impacts (elk causing damage to fences)? Did John find the Agreement? Read the agreement to know terms better. The gates at a min. should have reflectors (safety) and signage (WRA No Trespass).	Jennifer Bodensteiner	1
Oak Shores II (emergency road)	Not Started	Inspection of road and agreement review. Update gates and lock combinations. Trespassing occuring from Oak Shores and adjacent landowners.	Jennifer Bodensteiner	1
SLO fire safety council and Lease 3A.	In Progress	Receive project update from Dennis Javins on lease 3A as he was before the board about 1 year ago. Tom will get an update.	Tom Shepherd	1
Trespass Issue	Not Started	Property owner "Delgado" notified to stay-off Agency land and continues to trespass. Delgado is trepassing across Lease 4 (Soares) to access reservoir and will not stop. Need better means to enforce trespassing.	Jessell Fenley	1
Trespass Issue	Not Started	Agency "cross parcel" and trespass by Cal Shasta Boat Club members.	Jennifer Bodensteiner	1
Trespass Issue	Not Started	Tri-Counties illegal use, no (road) easement across Lease 3B relatd to Cullen site issue. Ask B. Buche.	Jennifer Bodensteiner	1
Trespass Issue	Under Review	Yates property near Lease 2 (Botts) and fence encroachment. Tom will check to see if the metal pickets are removed. Formal notification to be sent to Yates regarding correcting boundary line dispute.	Jennifer Bodensteiner	1
Overton road easement	Not Started	Request for road easement across Agency "cross" parcel. Ask B. Buche.	Jennifer Bodensteiner	1
Easement through Lease 3A	Not Started	Chimney Fire and property owners road easement from Cow Camp Road. Tom will provide name of owner and potential receive the agreement from him. If not, research and find.	Jennifer Bodensteiner	1
Hughes bridge easement	Not Started	Request for bridge easement across Agency parcel. Ask B. Buche.	Jennifer Bodensteiner	1
Potential new grazing leases (Batrums)	Not Started	Consider new grazing leases on Agency property at Naci Reservoir. Benefit would be fire suppression and increased revenue.	Jessell Fenley	1
Lauritson (roadway easement issue San Antonio north shore parcel	Completed	(Recently completed). The Stimson/Lauritson item has been resolved. The Stimson party finally agreed to grant the easement. The WRA will be filling a dismissal in the case next week.	Kelly Donlon	0
Bryant Canyon (8th Street, pipe easement request)	Completed	(Recently completed). The WRA BOS granted an easement to the developer (K. Slama) in March 2021 for the required drainage pipe (10' by 41' easement for installation of 12" diameter outlet pipe from retention basins). Applicant is pending final approval for discharge permit from USACE and Regional Board.	Manuel Saavedra	0
Bryant Canyon (Metz Road, bridge for community park)	Completed	(Recently completed). After working with the City of Soledad, the Cemetary district stated its opposition to the bridge project and access. Without the support of the Cemetary the project is not feasible and the initial request has been withdrawl.	Manuel Saavedra	0

Page 3 of 3

Monterey County

Board Report

Legistar File Number: WRAPA 21-025

June 04, 2021

Board of Supervisors Chambers

168 W. Alisal St., 1st Floor Salinas, CA 93901

Introduced: 5/27/2021

Version: 1

Set next meeting date and discuss future agenda items



Current Status: Draft Matter Type: WRA P&A Item

